# University of New England **DUNE: DigitalUNE**

All Theses And Dissertations

Theses and Dissertations

5-1-2016

## Assessing Transportation Management Associations (TMAs) In Rural Maine As An Approach To Increase Transportation Options

Vicki L. Rusbult University of New England

Follow this and additional works at: http://dune.une.edu/theses

Part of the <u>Educational Leadership Commons</u>, <u>Infrastructure Commons</u>, and the <u>Transportation</u> <u>Commons</u>

© 2016 Vicki Rusbult

#### Preferred Citation

Rusbult, Vicki L., "Assessing Transportation Management Associations (TMAs) In Rural Maine As An Approach To Increase Transportation Options" (2016). *All Theses And Dissertations*. 65. http://dune.une.edu/theses/65

This Dissertation is brought to you for free and open access by the Theses and Dissertations at DUNE: DigitalUNE. It has been accepted for inclusion in All Theses And Dissertations by an authorized administrator of DUNE: DigitalUNE. For more information, please contact bkenyon@une.edu.

## ASSESSING TRANSPORTATION MANAGEMENT ASSOCIATIONS (TMAs) IN RURAL MAINE AS AN APPROACH TO INCREASE TRANSPORTATION OPTIONS

By

Vicki L. Rusbult

BUA (University of Maine) 2008 MPA (University of Maine) 2010

#### A DISSERTATION

Presented to the Affiliated Faculty of

The College of Graduate and Professional Studies at the University of New England

In Partial Fulfillment of Requirements

For the degree of Doctor of Education

Portland & Biddeford, Maine

May 2016

Copyright by Vicki L. Rusbult 2016

#### ASSESSING TRANSPORTATION MANAGEMENT ASSOCIATIONS (TMAs) IN RURAL MAINE AS AN APPROACH TO INCREASE TRANSPORTATION OPTIONS

#### Abstract

Maine has the distinction of being a beautiful state defined by natural forests, crystalclear lakes, and a landscape unblemished by the usual marks of the densely populated regions of the U.S. The downside to this rural beauty is the remoteness of the sparsely populated communities which present challenges to residents who are unable to drive themselves to procure basic services because of physical, legal, or economic conditions.

Many rural residents are unable to rely on personally-owned, single driver vehicles. According to the Maine Department of Transportation Bureau of Planning (2013), the Eastern Maine region has an estimated unmet need (number of trips needed compared to available services) of 75% (Penobscot County) and 84% (Piscataquis County). The recent reduction in the cost of gasoline has made travel more affordable, but the reality is that many can't afford the purchase price tag, insurance, and/or maintenance costs to keep their vehicle on the road. The funding trends at the state and local levels will not support an expansion of public transportation to bring rural residents to hub centers for social services, healthcare, school, employment or other destination points. As noted in the Maine Department of Transportation Final Strategic Plan 2025 (2015, p. 13), federal funding in rural regions has remained at 2012 levels, and there is little local funding support for rural transit systems operating outside the more urban cities of Bangor and Ellsworth.

iii

A review of the literature on transportation challenges and barriers, particularly in respect to the sparsely populated region of Eastern Maine, provided the basis for this grounded theory qualitative study. A series of thirteen interviews were performed with study participants living in Eastern Maine who were characterized by one or more of the following traits: Low-income; elderly; medically restricted; student; and commuting worker. The results of the study provide insight into factors contributing to the gap, and whether a transportation management association (TMA) and rideshare boards represent reasonable solutions. The study also suggests methods to encourage use of alternative solutions to get Eastern Maine residents to their destinations. University of New England

Doctor of Education Educational Leadership

This dissertation was presented by

Vicki L. Rusbult

It was presented on May 9, 2016 and approved by:

Michelle Collay, Ph.D. Lead Advisor University of New England

Carol Holmquist, Ed.D, Secondary Advisor University of New England

#### ACKNOWLEDGEMENTS

I would like to thank my committee – Dr. Michelle Collay and Dr. Carol Holmquist – for their considerable guidance and support through my doctorate journey. Their direction helped me push through when I felt overwhelmed and helped me keep my goals in sight. I will always appreciate their encouragement. I would also like to thank my doctorate cohort partners for their support, as well as Cindy Lewis who transcribed my interviews.

I would be remiss to not thank my sisters Nancy and Cindy and their families for their support when I made the rather unexpected decision to pursue a doctorate of education. Rather than discourage me from taking on such a challenge, they were supportive and encouraging through the entire process. My mother Phyllis continues to be an inspiration as was my dad Rex, whom I dearly miss. I couldn't have asked for more supportive parents.

Finally, I want to thank my two sons, Willie and David. They are my rock and my inspiration, and I hope I have served as a model by showing that it's never too late to pursue your dreams.

CHAPTER 1: INTRODUCTION
Problem Statement
Purpose of Study
Research Questions
Research Question #1
Research Question #2
Research Question #3
Research Question #4
Conceptual Framework
Definitions
Assumptions10
Significance11
Conclusion
CHAPTER 2: LITERATURE REVIEW14
Transportation in Rural America14
Cost of Single Driver15
Land Use Patterns
Inclement Weather
Access to Education
Access to Work
Access to Healthcare

### **TABLE OF CONTENTS**

Demographics of Rural Maine	
Alternatives to Traditional Transportation	
Car Sharing	
Telecommuting	
Mobile Unit Healthcare Delivery	27
Rideshare/Cell Phone Networks	27
TMAs and Community Collaborations	
Collaborative Partnerships for Travel Planning	
Characteristics of the Successful TMA	
Behavioral Changes	
Conceptual Framework	
Conclusion	
CHAPTER 3: METHODOLOGY	
Setting	
Participants	41
Qualitative Data	
Analysis	44
Participant Rights/Confidentiality	45
Potential Limits of Study and Conflict of Interest	45
Pilot Study	46
CHAPTER 4: RESEARCH FINDINGS	47
Study Participant Interviews	49
Participant Profiles	

Analysis Methodology53
Common Themes
Theme #1: Relocation to Services Centers was Common
Theme #2: Fear of the Unknown and Personal Safety were Prohibitive
Factors
Theme #3: Lack of Awareness of Available Resources is a Barrier
Theme #4: Creation of an Environment to Promote Usage is Important60
Theme #5: Lack of Transportation has Caused Missed Healthcare
Appointments61
Theme #6: Substance Abuse Issues Can be a Prohibitive Factor
CHAPTER 5: CONCLUSIONS
Increased Visibility of TMA67
Increased Confidence with Safety Issues
Technology Challenges
Missed Healthcare Appointments70
Behavior Modification71
Methodology for Outreach
Concluding Observations
REFERENCES

### LIST OF FIGURES

Figure 1:	Rural TMA Characteristics	32
U		
Figure 2:	TIGER2 Grant Area	40

## LIST OF TABLES

Table 1:	Participant Profiles	53
Table 2:	Common Themes	55

#### **CHAPTER 1**

#### **INTRODUCTION**

Maine has long been recognized as one of the most beautiful states in America, with nearly 83% of the state comprised of pristine forests, hundreds of lakes, and an unblemished landscape. The commitment to sustain Maine's natural legacy is evidenced by the banishment of billboard signs along interstate and other highway systems that could mar the untarnished image. This beauty has earned Maine several monikers including "Vacationland" and "Maine – The Way Life Should Be", and demonstrates the state's reliance on the tourism industry that supports thousands of vacationers each year. Unfortunately, the very aspects that make Maine a desirable destination spot for vacationers characterize another facet of Maine. The state is extremely rural and sparsely populated, with the total population recorded with the 2010 census at 1,328,361, and the average density of its population recorded at 43 persons per square mile (2010 US Census). Maine's largest city of Portland proper has a population well below 100,000 residents.

Maine, like many other rural states, is populated by residents that are dependent on personally owned vehicles to travel independently to work, obtain an education, and for socialization, recreation, healthcare, and social service visits. There were 1,019,738 licensed drivers in the state in October 2012; 14.5 billion vehicle miles were traveled in 2010, with this number projected to increase another 15% by 2030 (TRIP, 2012). Based on population and other lifestyle trends, TRIP estimates that travel on Maine's roads and highways will increase another 15 percent by 2030.

It has become clear in recent years that this individualized approach to transportation will not be sustainable in the coming years (Simmons, 2013). While historically this approach to rural transportation may have allowed residents to access services, the increase in the cost of fuel

and declining presence of local jobs and services may necessitate a shift in transportation patterns and behaviors.

Many factors impinge upon rural residents' ability to transport themselves. According to Sharpiro, Pearlmutter & Schwartz, 2012, the high cost of fuel typically puts a strain on rural residents' household budgets and most economists predict that this trend will continue despite brief and temporary drops in the per gallon price (p. 93). The impact of burgeoning fuel needs in foreign countries like China and India was having a direct, negative impact on U.S. pockets, with the most vulnerable populations being those that must travel distances to get where they need to go (Shapiro, Pearlmutter & Schwartz, 2012, p. 92). A shift in the pricing of crude oil and related reduction in the average cost of gasoline during 2015 has reduced the financial strain on the average driver (U.S. Energy Information Administration, 2016). However, the fluctuating cost of fuel and its negative impact on household budgets is one factor that warrants investigation into alternate modes of affordable travel.

Another characteristic of Maine is the increasing proportion of Maine's senior population that is either independently choosing to remain in their homes and "thrive in place", or those who because of financial hardship are forced to remain in their historical homes (Colgan, 2006). Boober (2015) notes that Maine's current infrastructure of providing services and care to the elderly who are thriving in place is becoming unwieldy due to the expanding older population, and new strategies are needed to meet basic needs including transportation. As their ability to transport themselves decreases, the need for increased transportation options will continue to grow for the elderly to connect to healthcare services, social activities, shopping, and other services. Shifts in behaviors are also needed in a culture where many, in particular seniors, are reluctant to share hardships and prefer to compartmentalize transportation problems by hiding

them from others; there is a prevalent perception that sharing difficulties could put them at risk of losing their independence entirely (Hiskey, 2013). Hiskey notes that many seniors feel that admitting any type of hardship would bring unwanted notice.

The migration of workers from small communities to larger towns and cities has exploded over the last decade. The paper mill and manufacturing industries that once dominated the region have left the state taking well-paying jobs with them and leaving many communities without an industrial base. In early 2011, the *Bangor Daily News* reported that "a conservative think tank says Maine businesses have been losing more jobs on the average from closures than they created through startups between 1993 and 2007" and "on average, created 29,755 jobs per year from startups, but lost 30,359 jobs from closures." This industry shift has created displaced workers needing to drive longer distances to find employment. In 2014, this factor grew exponentially with the closure of additional paper mills resulting in the loss of more than 1,000 full-time jobs (BDN, 2014).

The migration of workers has not resulted in the relocation of these workers to more urban towns. Many rural residents are financially limited and are forced to remain in historical family homes; although the housing stock in rural communities continues to degrade, the cost to relocate (mortgage, taxes, insurance) is prohibitive (Maine State Housing Authority, 2008). These demographic and economic changes, combined with the limited funding available to expand traditional transportation modes, are driving the need for Maine to find more creative, affordable, and reliable solutions (Maine Statewide Strategic Transit Plan 2025, 2013).

#### **Problem Statement**

Because of the lack of public transit systems (bus, rail) beyond the borders of Maine's most populated communities, it is not surprising that one of the major challenges faced by rural

residents is access to reliable and affordable transportation (Maine Department of Transportation, 2010). The 2014 TIGER2 Planning Grant Report detailed that people living in rural areas, particularly low-income, elderly and the disabled, are more likely to lack reliable vehicles and live many miles from their work, health care, and social service providers (p. 3). This report also noted that Maine residents often face barriers to accessing higher education, social services, livable wage jobs, adequate housing, and other community supports (p. 4).

A rural-based Transportation Management Association (TMA), supported by the active participation of traditional and non-traditional community and regional partners, is seen as an effective model to close these gaps by connecting residents in need with viable transportation solutions.

#### **Purpose of the Study**

The purpose of this grounded theory qualitative study is to explore the perceptions of selected rural residents of Eastern Maine regarding their need for transportation and their willingness to utilize a TMA model to meet their needs.

The sample population was selected from five demographic populations, and the study focused on determining the study participants' perceptions, beliefs, and biases toward the use of a TMA in a rural setting to support connections with rides to healthcare-related services. Study participants included the elderly, disabled, students, commuters, and low-income persons who reside in rural communities now or in the past.

The findings of this study will contribute to the knowledge regarding rural residents' perceptions relating to the use of TMAs, in particular rideshare boards, and may be useful to those looking at the viability of implementing TMAs in Maine. These findings will be shared with transportation experts (local, regional, statewide and nationally),

service providers and community officials to help guide planning strategies for the coming decade.

#### **Research Questions**

Given the abundance of data as related in the Literature Review that indicates a need for increased transportation options for rural residents of Maine, in particular those living in Eastern Maine, four questions were identified to guide this research study:

 What were the perceptions of the study participants in rural Eastern Maine about the alternatives presented through the Transportation Management Association (TMA) including the rideshare board and volunteer systems?

2. Do study participants in rural Eastern Maine needing transportation consider TMA's a realistic solution for meeting their transportation demands?

3. Do study participants in rural Eastern Maine perceive that TMA options will increase their ability to keep medical appointments?

4. Do the participants indicate that they will be likely to access TMA options to keep medical appointments?

#### **Conceptual Framework**

The utility of TMAs in rural regions, and their impact on closing transportation gaps particularly for those who are elderly, disabled, or middle to lower income underserved populations, has been gaining momentum in rural states across the United States including Texas, Iowa, Vermont and New Hampshire to name just a few. Ferguson (2007) notes that TMAs have been used by a variety of stakeholders including employers, local government, childcare providers, social services, and others who have identified a need to connect people with rides. The comparative cost-effectiveness of TMAs, many of which are funded through

membership dues, has made them a viable alternative to federal and state-sanctioned services. The Vermont Agency of Transportation-supported Go!Vermont has embraced the concept of TMAs by offering carpool and vanpool services and access to other resources for people looking to reduce transportation costs and the environmental impact of driving.

TMA's may function through an established service-based organization, or as a stand alone umbrella transportation planning organization which would plan and partner with other organizations. A TMA also tracks specific goals, metrics, and desired community outcomes - for instance, the reduction of commute times, congestion and pollution or increased number of transit riders, parking available to visitors, and bike parking spaces.

A recent survey performed by the Penquis community action program in the Penobscot County region of Maine showed that 51% of respondents listed the lack of transportation as a major barrier to accessing healthcare services (Penquis, 2014). The interviews employed for this grounded theory qualitative research study focused on perceptions of the reliability of rides found through the TMA community rideshare board, identifying perceived barriers and methods for marketing the TMA by potential ride board users, and the utility for accessing healthcarerelated appointments and services. Study participants recruited for the interviews represented the broader needs identified in this study for employment, education, and other commute-related needs. The interviews will also serve as the venue to discuss the issue of missed healthcare appointments and the impact to the individual's overall health and well-being. Questions were crafted to investigate the value of a TMA for rural residents needing rides to connect to healthcare appointments and disease management services.

There are numerous organizations and national groups addressing transportation gaps that served as excellent resources for this research study. For example, the Association of State

Highway and Transportation Officials (AASHTO) created a tool for states to perform a selfassessment of transportation systems, and future socioeconomic and transportation drivers. This assessment and similar tools and resources were reported in the Literature Review and provide specific data about rural transportation challenges across the United States, and shaped the conceptual framework of this study.

Findings from this study will help deepen the understanding of the transportation needs of rural residents. The study will be shared with multiple audiences representing community leaders, service providers, educators, employers, state government representatives, and transportation professionals who are charged with finding real-world options for those in need of transportation solutions. Enhancing this study's Literature Review was the integration of findings into real world dialogs (referenced in the Literature Review) through collaborations with the State of Maine Department of Transportation, particularly through participation on the steering committee for Maine's Statewide Strategic Transit Plan 2025 and the federal MAP 21 funding plan to create Regional Transportation Planning Organizations to address transportation issues at the local level (U.S. Department of Transportation, 2014).

The conceptual framework for this study is based upon a platform of three primary constructs:

1. Identify the sociological factors that influence residents' health and well-being.

2. Explore the theory that individualized transportation via a self-owned automobile is unsustainable.

3. Build an alignment and collaboration between social agencies and resources to overcome barriers for rural transportation.

Understanding what influences the wide gaps of service in rural regions will be critical to guiding the evolution of TMAs as one solution to close gaps.

#### Definitions

The following concepts will be referenced throughout this research study:

<u>Community-Based Collaboration</u> is the partnership of public and private, traditional and non-traditional stakeholders who work together to address an identified problem and find solutions that benefit the communities where they live and work.

Eastern Maine refers to the region of Maine that encompasses Hancock, Penobscot, Piscataquis, Waldo, and Washington counties. This area of Maine is also referred to as Downeast Maine (Hancock, Washington, and Waldo counties) and portions of the Katahdin region (Penobscot and Pistacaquis counties).

<u>Ehealth</u> is an emerging field in healthcare that comprises principles from healthcare, technology, public health, and business. It presents as a model that promotes a state-of-mind and investment of a broad partnership to improve healthcare locally through communication and information technology (Eysenbach, 2001). It is particularly useful when managing chronic health conditions.

<u>Rideshare Board</u> refers to a repository posted on-line for connecting with volunteer and private or non-profit transportation options. Rides offered are posted as well as listings for persons needing rides. Users can access rideshare boards to connect for offering or seeking transportation.

<u>Ridesharing</u> is a description of riders who carpool to work, school and other destinations. Typically, expenses are shared and opportunities for sharing a ride are posted in a public site such as an on-line ride bulletin board.

<u>Rural</u> is described by the United States Department of Transportation (USDOT) as dispersed, isolated counties or regions with few or no major population centers of 5,000 residents or more. The regions are characterized by agriculture and natural resource-based economies, stabilized or declining populations, and facing challenges as a result of the dearth of public transportation.

<u>Telecommuting</u> is the use of the internet as an alternative to traditional transportation via an automobile. Using high speed connectivity, the user can work from home (home office), take long-distance on-line education courses, shop, bank, connect with healthcare providers, and perform other tasks without the need of a vehicle.

<u>Telemedicine</u> is a methodology employed using either the telephone, internet or shared networks for the purpose of providing consultation, chronic disease management, and education. The communication can occur between two health providers, a provider and chronic care manager, or the health professional and patient.

<u>Transportation Gaps</u> can be described as the recognition that people living in rural regions experience an inherent lack of options for getting where they need to go when they need to go. Current solutions only provide minimal relief.

<u>Transportation Management Association (TMA)</u> is described by the TDM Encyclopedia (2011) as a non-profit organization that provides advocacy, education, outreach, and links to transportation services in a particular area. The TMAs are often member-driven organizations that manage a wide range of services including ridesharing, carpooling and other options for non-car owning individuals.

<u>Transportation Solutions</u> can best be described as a roadmap developed as a result of a strategic planning process resulting in multiple and often innovative remedies and services.

<u>Thriving in Place</u> is a term describing retired seniors who want to retain their independence and remain in their homes and communities (also referred to as Aging in Place) (Partners for Livable Communities).

<u>Volunteer Ridesharing</u> is the term used to describe the provision of rides by volunteer drivers who receive no monetary compensation for the use of their private vehicle or the cost of gasoline. Rides are offered by volunteers for those who lack personal vehicles or have a disabling condition that prevents them from driving themselves (i.e., elderly, handicapped).

#### Assumptions

According to the 2009 National Household Travel Survey (NHTS), rural travelers tend to drive more often, are less likely to ride public transportation vehicles (primarily because of the lack of services), and drive vehicles that tend to be older and less fuel-efficient than their urban counterparts (NHTS, 2009). This holds true for drivers in Maine who historically prefer to travel alone in their private vehicles, even if those vehicles are less reliable or cost-effective. As noted in the TIGER2 feasibility study completed in February 2014, driver preference was linked to sole occupancy (Appendix A, p. 57). However, the TIGER2 study results reported an even greater number of respondents indicating that a lack of options often prohibited any travel for work, school, services, and social and recreation purposes (p. 22).

The TIGER2 feasibility study noted that there is the expectation that independentthinking Maine residents will be resistant, at least initially, to the options proposed by the TMA (e.g., rideshare, park & ride, cell phone networks, etc.) (p. 9). The six-month time frame from start-up to evaluation may be narrow to assess the true potential for change. This research will assess rural residents' assumptions about the benefits and risks of non-traditional transportation

options in an effort to determine the viability of a TMA.

The decision to assess the impact of gaps in transportation on healthcare management in this study was influenced by the observation that the lack of systemic and reliable transportation was a barrier for low-income, elderly, and handicapped persons. Accessing timely and ongoing chronic disease management support as well as preventive healthcare is critical for patients to maintain their health status while reducing costly emergency department visits and hospitalizations (Community Transportation Association of America 2014). Evaluating the impact of the TMA on healthcare management is the focus of this research, and was dependent on the study participants' willingness to participate in a survey that asked questions about their healthcare management and status. My role was to adequately promote and explain the importance of the research study to increase participation.

Based upon preliminary research performed in Eastern Maine through the TIGER2 feasibility trial, it is believed that engagement and commitment by community partners will continue as the TMA increases visibility. This commitment will be critical as the TMA is integrated into the lexicon of rural residents; confirmation of the TMA's usefulness and referrals to the TMA system will serve to encourage residents to use the agency as a resource.

#### Significance

The recent TIGER2 Planning Grant Report noted that many rural residents of Maine are challenged on a daily basis to connect with viable transportation to work, school, healthcare appointments, and other services -- even traveling to the grocery store (p. 4). With the current structure of national and state funding mechanisms for transportation-related needs, the TIGER2 report states that the identification of new funds to expand public transportation venues seems remote (p. 9). It is critical that Maine transportation agency leaders think outside the box to find

realistic solutions to close the current gaps (Maine Statewide Strategic Transit Plan 2025, 2013). Finally, findings summarized in the TIGER2 Planning Grant, 2014 conclude that these solutions will need to be found locally to assure that options identified are realistic for the region (p. 4).

Finding those answers will require that local stakeholders take the lead. To that end, this study explored the perceptions of selected rural residents of Eastern Maine regarding their insights about their need for transportation and their willingness to utilize a TMA model to meet their needs.

The transformational leader (or possibly the leader among a team of leaders comprising the team) can provide the necessary support to guide the collaborative partners in identifying and implementing new options. Solutions will need to be realistic and framed by the findings of this and other research that takes the biases and preferences of users into consideration; the transformative leader can direct efforts by identifying compatible programs in other regions and assure that the final plan has been sufficiently researched to warrant implementation. Finally, the transformative leader will be the voice to remind partners of equitable and fair consideration for all rural residents in need.

#### Conclusion

Chapter 1 identifies the problem of transportation gaps in the remote rural regions of Eastern Maine that is exacerbated by barriers that have widened these gaps. In recent years, service providers have experienced a shift by state and federal administrations that historically intervened to identify resolutions and provide sufficient funding to minimize transportation gaps (Maine Statewide Strategic Transit Plan 2025, 2013). Instead, the onus is now falling to community-based, non-profit, and grassroots-driven organizations and stakeholders. In addition to the statement of problem, chapter 1 identified the purpose of the study, the research questions

to be answered, the conceptual framework, the assumptions that impacted the completion of the study, and the veracity of the findings.

Chapter 2 provides an in-depth literature review to describe the current gaps, examples of solutions, and an introduction of the TMA. Chapter 3 provides a review of the methodology utilized in the study. Chapter 4 presents data and Chapter 5 presents conclusions and recommendations.

#### **CHAPTER 2**

#### LITERATURE REVIEW

The literature review describes and assesses the components unique to rural transportation barriers and areas of concern:

<u>Transportation in Rural America</u>– identification of socioeconomic status, and transportation characteristics and impacts

<u>Demographics of Rural Maine</u> – review of barriers and characteristics which impact rural mobility in Maine

<u>Alternatives to Traditional Transportation</u> – introduction to multimodal options beyond public transportation

<u>Transportation Management Associations (TMAs) and Community Collaborations</u> – definition of one emerging solution to rural gaps and compelling characteristics that shape TMAs

<u>Characteristics of the Successful TMA</u> – recommendations from the literature and the critical components of interest

<u>Behavioral Modification</u> – review of factors that support changes to behavior that influence increased usage of TMAs.

Chapter 2 reviews rural development concerns contributing to or influenced by the lack of transportation alternatives for rural residents, and reviews options that have been identified as potential solutions to close the gaps and increase opportunities for residents to get where they need to go. This chapter describes the characteristics of transportation in rural America and rural Maine, most notably the historical prevalence and unsustainability of single-driver vehicles and land use patterns that discourage shared ride alternatives. The results of these prohibitive qualities along with Maine's climate and periods of harsh weather have a direct impact on access to education, socialization and cultural inclusion. An overview of the lack of transportation's impact on access to healthcare and the hidden health costs of this gap will be presented. Following a demographic portrait of Eastern Maine, the chapter will shift to a discussion of alternatives to traditional modes of transport, the value of TMAs and community collaborations, and the impact of addressing behavioral patterns to enhance the acceptance of alternative solutions to close transportation gaps for those in need. Finally, a review of the organizational characteristics inherent to TMAs will also be identified. The literature review provides the basis for this research study and quantifies why this research is needed particularly in rural Eastern Maine.

#### **Transportation in Rural America**

Accessible and affordable transportation has a critical role in determining the livability in communities and quality of life for those living in rural regions of the county. Travel behaviors and barriers to travel for even basic needs vary greatly between urban residents and those residing in remote rural communities where public services are sparse (Rural Transit Fact Book, 2014). As noted in this section, the demographic backdrop of rural Maine residents, the costs associated with single-driver transportation, land use patterns, and other barriers all negatively impact rural residents' transportation capabilities. These barriers directly influence residents' ability to access education, work, healthcare and other basic needs for emotional, physical and socio-economic well-being.

#### **Cost of Single-Driver Transportation.**

Maine's historical reliance on and prevalence of single-driver transportation is unsustainable. Brown & Stommes (2004) note that disadvantaged, low income rural households are three times more likely to lack a working vehicle than higher income homes. This

disadvantage extends to preventing livable wage employment; the lack of a vehicle and growing need to travel greater distances to employment is often an insurmountable barrier to securing employment and relegates some rural residents to a life of continued generational poverty. According to a report from the USDA (Cromartie, 2014), rural populations as a whole are declining with most rural counties across the United States beginning to experience losses since 2010. The recent trend of declining population in rural communities means that increased public transportation will not be an option.

The high cost of gasoline in recent years served as a catalyst for more rural residents experiencing difficulty getting where they need to go. Johnson (2011) further states that rising fuel costs will ultimately change how the average citizen travels and will emphasize their need to access public transportation and other similar options. Despite current national gasoline prices averaging a low \$2.05 a gallon, many experts predict that gasoline prices will begin to hike in late 2015 and 2016, and could reach \$5 a gallon by the end of this decade (Loveless 2015). This prediction reaffirms the need to identify alternative solutions in the long-term for one-driver vehicles.

#### Land Use Patterns.

Litman (2005, 2006) contends that historical land use patterns and current transportation systems limit improvements supporting an expansion of options for disadvantaged travelers as they demonstrate a preference for single-driver automobile travel rather than alternative modes. He notes that current funding practices are skewed to favor increased automobile dependency by demonstrating a preference for parking and roadway facilities over more cost-effective alternatives. Smart growth development is successful when solutions identified can provide benefits that support more diverse and efficient alternatives (VTPI 2014). Land use planning

includes locating common destinations close together to increase accessibility and transport diversity, but common usage is devoted to more urban, heavier populated locales rather than to support efficiency for longer distance commuters. Litman (2014) suggests that planning should include distribution of the management of demand to increase capacity equally with focus ascribed to distinguishing the economic, social and environmental benefits.

#### **Inclement Weather.**

A factor that merits consideration is the inclement weather in Maine. On an average of 6 months each year, Maine is characterized by unpredictable and often severe weather patterns. For the rural resident, this often means that traveling distances is out of the question. Frigid weather and snow can present difficulties for travel as roads are often closed and transportation agencies (including healthcare transporters) discontinue services until the weather clears (Goins, Williams, Carter, Spencer & Solovieva, 2005).

#### Access to Education.

The absence of a reliable vehicle restricts many from attending higher education institutions. Fletcher et al. (2010) note that the lack of reliable transportation is a barrier and is evident at numerous points of need including higher education and employment. The lack of post-high school education is a factor that can inhibit one's ability to obtain work in a higher pay range. Higher wages lead to an improved quality of life, with benefits noted for both the individual and society as a whole. These benefits include the ability to negotiate higher incomes, greater job choices, satisfaction, and security (Hill 2015). Barriers to higher education have resulted in many rural Maine residents lacking higher degrees beyond high school; for example, 16.2% of residents aged 25 and older in Piscataquis County attained a bachelor's degree or higher compared to 28.5% in the U.S (U.S. Census, 2010). Even the advent of on-line, distance

learning is does not always address rural education access because of the lack of high-speed internet connectivity. Many rural communities still rely on dial-up services which do not support the capacity needs for on-line education platforms. This "digital divide" represents yet another barrier to rural students that puts them at an educational and economic disadvantage (Speedmatters, 2008).

#### Access to Work.

In Eastern Maine (and similarly in other rural areas across the country), closure of historical manufacturing companies and factories are forcing rural residents to travel further for employment. In Eastern Maine, the closure of the paper mills has resulted in the loss of more than 2,000 livable wage jobs. For the first time in over 120 years, there are no active paper mills in Eastern Maine. Because these companies were often the heart of small rural Maine communities, displaced workers needing to search for new employment have had to look further away to larger hub centers for jobs. Often these jobs are at a reduced pay level with fewer employee benefits, resulting in a reduction in the overall household budget (Camoin 2015). For those without vehicles, securing transportation to employment is challenging.

#### Social Integration and Cultural Inclusion

Community activity and socialization has a significant impact on the quality of life of seniors aging in place (Shergold, Parkhurst & Musselwhite, 2011). Access to cultural venues serves to increase one's sense of community, and for older citizens can be an important socialization channel. Isolation often results in a withdrawal from family, friends, and community. A recent Gallup-Healthways Solutions poll found that people of all ages but in particular seniors, showed great benefit to their health and well-being (i.e., reduction in stress, loneliness, depression impacting basic vital signs) by even a few hours of socialization with

family and friends on a routine basis (Gallup-Healthways, 2011). In rural areas, this often entails travel to central hub locations for social interactions.

#### Access to Healthcare.

Mitton, Dionne, Masucci, Wong & Law (2010) report that, although Canada requires that all citizens have reasonable access to healthcare, the reality is that many in remote rural regions of northern Canada are at an extreme disadvantage. This lack of access has resulted in a diminished overall health status of these residents as they lack the ability to manage chronic illness and receive timely care for acute episodes. Their research concludes that a multi-approach solution is warranted and includes the use of telemedicine and e-health (public health education and outreach) platforms.

The delivery of healthcare via the internet is growing in acceptance, particularly for patients who must manage chronic conditions such as diabetes, cardiac disease and more. The results of their research suggested that the reduction in the number of medical evacuations was substantial. This type of "intelligent" transportation is combined with more traditional methods of delivery to enhance the overall quality of care and increase access to health care.

In a study performed in six West Virginia rural communities by Goins, William, Carter, Spencer & Solovieva (2005), it was concluded that inadequate transportation was a contributing factor to poor health management. Contributing barriers included extended travel for specialty care, limited medical transportation programs, limited public transportation, and inclement weather.

#### Hidden Health Costs of Transportation

Benmar (2010) looks at another side of transportation and health impacts – those caused by a continued reliance on fossil fuels and high volume automobile usage. She notes that the

negative outcomes from our obsession with vehicle usage has led to an increase in the exposure to air pollution, bodily harm resulting from traffic crashes, and decreased opportunities or desire for physical activity (i.e., walking to destination rather than driving). Benmar points out that these factors have resulted in increased costs associated with obesity, pulmonary illnesses, cardiac issues, and injuries from crashes. Benmar also notes that "health impacts and costs have typically not been considered in the transportation policy, planning and the funding decisionmaking process." Integration of public health professionals as a non-traditional partner to transportation planning is critical to reduce the negative impacts.

The recent report on hidden costs of transportation published jointly by the Urban Design 4 Health, Inc. and the American Public Health Association (2012) confirms Benmar's conclusion and reports that the added annual costs to obesity (approximately \$395 per year per person), air pollution (\$50-\$80 billion), and traffic crashes (\$180 billion) justify a shift in transportation planning to consider health impacts and costs. They recommend that future strategies should also include a national set of health-related policy objectives to help drive funding to projects supporting healthy communities and active alternative to traditional transportation (i.e., walking and biking).

The Urban Design 4 Health, Inc., the American Public Health Association, and Benmar support a relationship between creative solutions to close gaps, and recommendations that include safe and logical routes that promote mobility through physical activity (Ferguson, 2007). While the attainment of safe transportation and mobility are critical, the reality remains that most rural communities have limited opportunities for walking or biking as viable travel solutions because of the great distances that must be covered to reach destinations.

Vrabec discusses the various issues that exemplify seniors living in rural communities. These include: 1) more individuals are living at the poverty level or at lower incomes; and 2) lack of ready access to healthcare results in higher incidences of unmanaged chronic illnesses (e.g., hypertension, cerebral vascular disease, cardiovascular disease, pulmonary disease, obesity, etc.). The trends of declining rurally-based health centers (primarily due to financial constraints and inadequate reimbursement for patient visits), a lack of awareness about existing services, decreased availability of healthcare providers, and the geographic dispersion of rural residents (Vrabec, 1995) all lead to the conclusion that increased reliable transportation is needed for this and other segments of the rural population.

The author points out that the lack of rural services and transportation often forces elders to move to urban settings and institutionalized housing. Among Vrabec's proposed solutions are a restructuring of the healthcare reimbursement system to possibly use a "resource-based relative value scale" (increased payments and provider incentives to practice in rural settings), increase the range and number of healthcare providers by modifying the state licensure laws, and increased provision of supportive services. Increased transportation options have a role in this list of solutions and could provide a more cost-effective alternative.

For the disabled person living in rural towns, an increased reliance on automobiles rather than public transportation or transportation services has been reported (Deka, 2014). For those lacking family members or friends to provide this service, options are limited. Deka also notes that barriers for the disabled will continue to increase due in part to the increasing number of seniors (extended longevity because of medical advances), the growing number of single households, and the increasing demand for scarce and often expensive transit services.

The barriers to adequate transportation and subsequent outcomes noted including decreased access to education, livable wage employment, healthcare services and socialization directly relate to the socio-economic conditions identified in Eastern Maine. Although the average cost of gasoline fluctuates, the reality is that the cost per gallon will rise in the coming years. This expense, coupled with historical planning and land use transportation patterns favoring the single-driver model, and Maine's harsh and often unpredictable climate continues to influence access to essential services and the Eastern Maine residents' quality of life. This discussion provides the foundation for the importance of this research study – identifying viable solutions to gaps in transportation.

#### **Demographics of Rural Maine**

Because of inherent geographic, psychological, financial, and social barriers, people living in rural areas of Maine are more likely to lack transportation, live many miles from their work, health care and social service providers, and are less likely to easily access higher education, social services, livable wage jobs, adequate housing and other community supports (Kolodinsky, 2013).

Many rural residents are "trapped" in place because of the unaffordability of housing in more urban locations; although the housing stock in rural communities continues to degrade, the cost to relocate (i.e., mortgage, taxes, insurance) is prohibitive (Maine State Housing Authority, 2008).

The Healthy Maine 2010 study reported that rural counties in Maine also tend to have older populations (anywhere from 13-14% of population in Penobscot, and 17% in Piscataquis) as well as higher rates of poverty and lower median incomes. These and other rural-specific factors contribute to the barriers and circumstances that result in gaps in transportation and

diminished access to services and employment. As described in the 2010 Healthy Maine report, the issues and characteristics related to the barriers have a negative impact on all aspects of life. The incidence of acute and chronic medical conditions and restricted access to healthcare are high and some rural residents are burdened by the inability to travel distances for healthcare. This leads to poorer management of chronic illnesses and missed appointments.

Fletcher, Garasky, Jensen & Nielsen (2010) note that reliable and affordable transportation is a key mechanism to enhancing economic outcomes for low-income families, most significantly for those struggling to manage employment and daily needs. Relevant variables include a lack of driving skills, high costs of insurance, maintenance, and repairs of vehicles, lack of access to consumer credit, and inability to obtain a driver's license – all leading to an overall sense of stress and reduced decision-making abilities.

A Maine State Plan on Aging (2012) notes that Maine is the oldest state in the nation as well as the most rural, reporting 15.9% of Mainers are age 65 and older. The 2010 Census Bureau projects that by 2030, Mainers age 65 and over will constitute 26.5% of the state's population representing a 12.1% increase. The median age has already been increasing, with a median age reported in 2013 at 43.9 (Bangor Daily News, 2014). The inability to drive is an obvious problem experienced by the aging. Accompanying this growth in the elderly is the expected reduction of those 18 and younger (projected reduction from 23.6% to 18.1%).

Finally, workers living in rural regions consistently travel longer distances when compared to those living in more urban areas such as Bangor, Augusta or Portland Maine. On average, the rural commuter drives for a mean travel time of 25 minutes or more one way compared to more urban-based travelers (20 minutes or less one way) (American Community Survey Reports, 2009).

Prior studies have been performed in Maine to accurately assess barriers to transportation and needs such as the Eastern Maine Transportation Collaborative (United Way of Eastern Maine, 2004). This study investigated the challenges and barriers seniors face when accessing chronic care medical services. A complementary study performed by the University of Maine Center on Aging (2005) expanded the research to determine current gaps and services. Both concluded that the lack of viable transportation was a critical barrier to effective disease management and overall health. These studies and others confirm that barriers are related to advanced age, lower income levels, lower attainment of education beyond high school, and a lack of public transit options.

Along with these barriers, the concerns related to personal safety are also important factors influencing the utilization of alternatives to traditional transportation. Beecroft and Pangbourne (2014) note that ingrained negative perceptions related to public and non-traditional transportation often deter individuals from its use. They define personal security as factors related to security, safety, and confidence (e.g., confidence in the mode of transportation selected).

These studies and others confirm that barriers are related to advanced age, lower income levels, lower attainment of education beyond high school, and lack of public transit options. For Maine, the additional obstacles of longer driving times for employment and behavioral biases add another layer of difficulty. The demographics described contribute to the systemic adversities experienced by rural residents lacking reliable transportation.

#### **Alternatives to Traditional Transportation**

Maine residents, like those in many other states, are dependent on personally owned vehicles to travel for work, health, education, social and all other necessary trips. Corless (2011)

argues that the continued increase of gasoline prices will be accepted by Americans as long as a direct correlation between transportation and improving their lot in life (more secure, convenient and affordable) is clear. Therefore, he contends that the federal government must participate in planning to invest in an array of new options so that the high costs of gas do not adversely affect the average citizen's budget. Although the cost of gasoline has declined in 2015, it is generally accepted that prices will spike closer to \$5.00 per gallon by 2020 (Loveless 2015).

Over the last few decades, efforts by transportation and service providers have begun to generate new creative alternatives to traditional modes of transportation. Public transit such as public buses and service-oriented buses will still be needed, but the cost to provide these services prohibits an expansion to meet growing needs (Corless, 2011). Some of these alternatives are described; some are applicable to rural settings while other options are limited to more heavily populated cities.

#### **Car-Sharing.**

The use of Zipcars has been gaining popularity as an alternative for those without a personal vehicle. Zhou (2012) discusses the value of university employee car-sharers in Los Angeles. He notes that this alternative transportation mode not only serves as a viable and cost-effective solution but also has a positive impact on greenhouse emissions (reduced rider volume on campus) and increases the mobility of the carless employee. Although this is an option worth considering in more populated communities, its value diminishes for rural communities. Many residents not only face financial barriers to Zipcar program membership, but the practicality of access presents another roadblock – many live in homes so rural that they would need rides to pick up and drop off the Zipcar. The University of Maine terminated the Zipcar program after
determining that poor usage, lack of availability, and poor acceptance resulted in under-utilization of the program to meet operating needs (The Maine Campus, 2008).

Zhou noted that the median income of participants of the Zipcar program was \$30,000-\$40,000; Maine median income is lower than the national average and even lower for those counties more rural in nature. The median household income for residents of Piscataquis County (total population 17,535) is \$36,606 compared to the state average of \$48,219, and more than 17.7% live below the poverty level (U.S. Census 2010).

#### **Telecommuting.**

As noted earlier, the use of telecommuting is also viewed as an effective model to eliminating barriers in rural transportation. The Portland Plan (Oregon) is a collaborative project promoted by the City of Portland and comprised of a partnership of various public and private partners. The goals of this plan focus on four priority areas – prosperity, education, health, and equity. Transportation has been a priority of this partnership, and their plan's identification of technology as a way to improve transportation for education, business (particularly home offices), socialization, and other services (for example, on-line banking) supports the TIGER2 determination that increased high-speed connectivity will open new doors for home-bound or low income residents. Their end-goals include reducing the number of miles traveled for care, increasing affordability of high speed internet access, promoting telecommuting, and increasing internet use for government services (Portland Plan, 2014).

Although seemingly a viable alternative, in rural regions of Maine broadband connectivity is poor with some areas still reliant on dial-up services as their only means of telecommuting. Until broadband connectivity is upgraded, travel through telecommunication portals is limited in Eastern Maine (Speedmatters, 2008).

#### Mobile Unit Healthcare Delivery.

Mobile units for healthcare delivery have been tried in rural communities as an alternative to permanently structured clinics that require the patient to travel for care. Services have ranged from annual and well-patient physical examinations, oral health examinations (including x-rays, fluoride treatments, and fillings), and mental health services. Most agree that the benefits of bringing healthcare to the community are great and ultimately reduce the need for emergency care while increasing the well-being of patients. The downside of mobile units as a venue for healthcare is the extreme cost of services – purchase of the specially equipped van, insurance, maintenance, staffing and associated per diem costs, and gas can be prohibitive and unsustainable. These costs often inhibit program longevity; on average, mobile clinics cost anywhere from \$300,000-\$375,000 annually. Most rely on donations and grants to sustain operations, and often face discontinuation when these cash streams end (Srinivasan, 2015).

#### **Rideshare/Cell Phone Networks.**

Sharing rides is another alternative that has gained popularity in recent years. The Upper Valley Rideshare (www.uppervalleyrideshare.com) in New Hampshire and Vermont is an example of a grassroots program that helps commuters connect with drivers for recurring commutes (i.e., work, school), one-time appointments, and rides to special events. A defining characteristic of ridesharing is that typically the driver and passenger are unacquainted and are meeting solely for the purpose of sharing transportation to a specific destination.

A methodology employed on a limited basis in Maine is the cell phone network – a small group of neighbors are connected through cell phones, with calls circulated to the phone list when trips are being made to service centers, grocery stores, the mall, and other destinations. The network provides advance notification when trips are scheduled for ridesharing or, in the

case of a grocery store trip, shopping lists can be shared allowing the driver to pick up goods and deliver to the consumer (particularly useful for house-bound and elderly citizens) (TIGER2 Planning Grant Report, 2012).

Despite the existence of alternatives to traditional transportation, evaluating their applicability to Eastern Maine has resulted in the elimination of car-sharing such as Zipcars and mobile unit services delivery as viable options to replace the single-car model. Telecommuting is complicated by the lack of the needed infrastructure; dial-up services are still prevalent in many corners of rural Eastern Maine. Mobile units, although attractive as a service provider vehicle, is cost-prohibitive. Finally, Uber and Lyft merit a mention, although their usefulness in rural settings (similar to taxicabs) is not likely. A majority of small rural communities already lack taxi services due to low ridership, and the import of Uber would likely not be cost-effective because of low ridership, long drives between fares, and overall trip revenue too small to cover gas and vehicle upkeep (Hawkins, 2015).

One notable exception is the application of rideshare opportunities, notably through TMAs, as a potential alternative in rural Eastern Maine.

## **TMAs and Community Collaborations**

A final option identified as a viable solution to reduce transportation barriers is the use of TMAs. Often, the mission and operations of these TMAs are community-based partnerships comprised of traditional and non-traditional partners. The value of this repurposed model as a solution to rural transportation gaps is gaining momentum as a viable solution that is demand-driven.

Initially launched in the 1970's, their popularity waned until the last decade when the benefit of a locally-managed organization was noted to be advantageous for efforts to control or

reduce emissions and traffic congestion in metropolitan settings. However, it wasn't until the last few years that the value of a TMA in rural settings was recognized. Ferguson (2007) reports that between 1990 and 2002, a total of 249 TMAs were in operation, with 231 situated in large urban centers and the remaining serving rural and small urban areas. Of the 18 rural TMAs, he noted that 13 were still in operation, a fact that demonstrated that smaller TMAs were more likely to succeed when compared to the large counterparts (Ferguson, p. 6). TMAs have also shifted focus to providing direct services to commuters, with less emphasis on advocacy (Ferguson, p. 24).

Ferguson, Ross & Meyer (1992) discussed the critical components of a TMA including their organizational structure (public-private partnerships were a key component to early TMAs), implementation including investment from communities, funding needs, staffing needs, and responsibilities, and the need for an evaluation of the early phases to determine if program modifications were needed. Ferguson (2007) updated his original paper to include case studies and a review of eight national TMA surveys performed between 1989 and 2003. Following the lull in TMA's use in the 1990's, they have again emerged in popularity in the 2000's by offering a wider variety of products and services, more financial security, and as viable organizational models to address changes in transportation due to increasing fuel prices. His review of survey results confirmed the decline of TMAs in the late 1980's and 1990's, with results showing that many were disbanded. Of interest is the identification of TMA dues as one of the more significant fiscal drivers to long-term success. It was also interesting to note that many TMA's did not include performance evaluation, thereby decreasing their ability to implement change or fixes to maintain their effectiveness (Ferguson, 2007).

The on-line TDM Encyclopedia on Transportation Management Associations (2014) provides an overview of the many elements and considerations when forming a TMA. Among

the components reviewed including implementation, benefits and costs, impacts to travel, equity impacts and regional application, TMAs were noted to have potential benefit for communities (including rural regions). This resource also provides an identification of the many solutions that TMAs can address such as ridesharing, telework, access management, shuttle services and more. There are many TMAs currently in operation that serve as resources for developing similar models in Maine. These include the New England models of Upper Valley TMA (Vermont and New Hampshire collaboration), Go Addison County (Vermont), and CATMA (Burlington VT).

The Upper Valley TMA has a membership base of private employers, transit providers, and town and regional planners. This model supports various service providers, workplace and employer connections, advocating on behalf of transit providers, and outreach to potential commuters (http://www.uppervalleyrideshare.com/). The Go Addison County model in Vermont works with a variety of partners to advocate for increased transportation efficiencies and support collaborations with businesses, public agencies, employee transportation networks, rideshare resources, transit and shuttle operations, and other partners in the region to increase travel options. The Campus Area Transportation Management Association (CATMA) is another example of a New England TMA model that, in addition to traditional roles, works actively to improve air quality and address parking demand (CATMA 2013).

#### **Collaborative Partnerships for Transportation Planning**

Rangarajan, Ziemer, and Long (2009) conducted a research study looking at the value of public and private partnerships to foster economic development activities and address infrastructure deficiencies. Specifically, their research focused on a presentation of two case studies that involved public-private partnerships. The first case study reviewed a transportation project that would connect rural towns via the implementation of a ferry service. The case was

made that although increased value was not prevalent for development projects, there was clear value in engaging partners for engineering projects such as public transit, improvement of roads and bridges, and evaluation of opportunities for innovative and technology-based systems management. The synergies recognized by the partnering of community groups and agencies with private enterprise aligned well with the goal of producing creative solutions and rapid engagement of same.

Deloitte Development LLC confirms these conclusions by stating that public-private partnerships (PPP) are invaluable to addressing numerous infrastructure needs including transportation, education, housing, waste and water, defense, and prisons. Because of the complexities and cost associated with governmental intervention, PPPs can play a critical role in closing the gap to identify infrastructure solutions (Deloitte, 2007).

In Eastern Maine, the formation of public-private partnerships has been gaining momentum as alternatives to historical intervention resulting from solutions by State or federal entities. The Eastern Maine TMA, launched in 2013, is one example of a successful partnership of community collaborations (<u>www.emdc.org/GettingThereMaine</u>). These grassroots organizations are a viable answer to the problem of providing alternative transportation that is accessible and affordable.

#### **Characteristics of the Successful Transportation Management Association (TMA)**

An evaluation of existing models proved useful as the Eastern Maine TMA was implemented to assure that all applicable components were addressed and subsequently evaluated for impact. Morris (2006) discusses the steps needed to adopt new or modified transportation systems management models that address the more complex issues of growth, congestion, safety, and other transit considerations. At the heart of his recommendations is a well-thought out and clear marketing plan to increase awareness and community recognition. Morris purports that

communication portfolios, strategies, and the "push versus pull" process is useful as the new TMA is created and the research is conducted to assess its utility.

Instructional "tools" that assist with the creation of new TMAs and identifies the critical component leading to a successful enterprise are available. The TDM Encyclopedia is one such guide that identifies those pieces to be considered when forming the new community-based organization including access management, marketing and promotion, rideshare matching, and telecommunication support to name a few (TDM, 2011). A visual representation of the rural TMA is provided in Figure 1.

Figure 1

## **Rural TMA Characteristics**



In Eastern Maine, the characteristics were assessed and tools were applied to support the development of the Eastern Maine TMA and subsequent rideshare board a focus on marketing the resource and managing the assets are critical pieces to assure its success.

## **Behavioral Changes**

Providing options for transportation will have limited value unless those needing rides are willing to use TMAs and local rideshare boards, and forego the preference to drive independently. For those unwilling to share their lack of transportation or access less conventional options, rideshare boards and volunteer rideshares may be a hard sell. Acceptance of this alternative will require a change in attitude that needs to be based in education and exposure to dispel misunderstandings or fear. Harvard Women's Health Watch (2012) notes that change is a process that requires multiple stages including pre-contemplation, contemplation, preparation, action, and maintenance.

Changes in behavior can also be influenced by life events or external pressures (Littell, 2002). In the case of utilization of TMAs, this could be the loss of vehicle ownership, aging, incapacitating medical conditions, or prohibitive cost. To this point, Schwarzbach (2013) reported that increasing fuel costs can be a catalyst to behavior modification. She notes that, for those lacking sufficient disposable income to absorb rising fuel prices, finding affordable alternatives to single occupant vehicles becomes more palatable but still requires education and a degree of persuasion to shift from thinking to doing.

One approach to modify rural residents' transportation behaviors is to identify the pros and cons, examining the barriers that inhibit utilization of rideshare boards and methods to bypass them (Harvard Women's Health Watch, 2012). This approach has been applied to modification of health behaviors, but is applicable to ride seekers.

Assessing the needs of a rural region, assets and resources available to enhance options, building the collaborative partnerships, and creating plans to impact behavior patterns of potential users are critical to the success of the alternative to traditional transportation.

#### **Conceptual Framework**

The literature review has clarified the characteristics of the Maine rural driver and rider, describing a population dominated by the preponderance of older, less educated, lower to middle income people who lack options to travel to meet their needs. Compounding these factors are the inherent barriers that characterize living in rural communities – the lack of public transportation, high costs associated with maintaining personal vehicles, and the lack of readily accessible support systems. In a regional transportation survey, it was noted that preferences and behavior were directly influenced by the need for reliable, stress-free, and productive transport (Popuri, Proussaloglou, Ayvalik, Koppelman & Lee, 2011). Clarifying attitudes and biases will be key constructs to the implementation of a viable alternative to single-driver vehicles and public transportation.

The TMA builds on a framework of community collaboration and encourages ownership of the solution at the local level by invested stakeholders. This study assessed these components to consider the TMA as one possible solution for remote and rural areas of Maine.

# Conclusion

The literature review articulated the problems and demographics that contribute to barriers associated with the lack of transportation in rural America, described the demographics of rural Maine residents and communities that negatively impact mobility, and presented alternatives to traditional modes of travel. The chapter also introduced TMAs and the need for effective community stakeholder collaboration to sustain the successful TMA; the characteristics of a successful TMA were described. Finally, the literature addressed the need to consider behavioral patterns that will determine whether a TMA s considered to be viable.

As a result, this literature review effectively created the framework to guide this study's research. Assessing the value of the TMA and the likelihood of utilizing its alternatives by "Mainers" was the core of this research study.

#### **CHAPTER 3**

## METHODOLOGY

This research study focused on one option to address transportation gaps in rural Maine – the rural Transportation Management Association (TMA). Central to this study was the documentation of the beliefs and preferences of rural residents living in a representative subset of Eastern Maine communities. Interview questions were posed to participants for the purpose of assessing perceived barriers to use of a rural community rideshare board. Resident input was gathered regarding the residents' perceptions of the TMA as a solution to gaps in transportation. The purpose of this grounded theory qualitative study was to explore the perceptions and biases of persons living in rural Eastern Maine about the need for alternatives to traditional public transportation and their willingness to use these options.

The research used a grounded theory qualitative approach to seek and analyze input from a diverse cohort of study participants who have a shared common experience (Creswell 2013, p. 83). For this research, the shared common experience was the lack of transportation options. Participant input was the basis for proposing an action or theory of a process that would incorporate study participants' perceptions and preferences (Bloomberg, 2012, p. 137). Because this methodology was grounded in the experiences and input from study participants, the end result provided data that will be a framework for further study and application (Bloomberg, 2012, p. 33).

The study participants were recruited from a varied sampling of service provider agencies including social services, elder, healthcare, and education for a theoretical sampling. As noted by Glaser and Strauss (1967), theoretical sampling is the process of qualitative data collection where i jointly collect codes and analyze data to support the emergence of a theoretical idea(s).

For this study, a theory was researched to determine if one could predict and explain behavior on whether rideshare boards and TMAs would be viable alternatives to single-passenger vehicles. Considering the emerging theory as the interviews progressed and channeling observations from prior interviews helped to provide a clearer picture to subsequent interviews and helped frame the conclusions reached.

Thus, the interview questions were designed to discover participant preferences, biases, beliefs, and experiences that personally influenced their access to transportation. Bloomberg indicates that the selection of individuals who can contribute to the evolving theory constitute the theoretical sampling to build the case for the research (p. 102). The selection of a diverse group of study participants representing the elderly, physically and/or mentally handicapped, commuters needing rides to work, low-income persons needing rides to a variety of destinations, and students ensured representation of those identified in the initial TIGER2 study as most in need of transportation options (2013 TIGER2 Report).

Differentiating grounded theory from other qualitative methodologies is the focus on collecting data to build a theory (Merriam, 2009). Merriam stresses that grounded theory supports research that is based upon real-world, everyday conversations with and opinions from study participants.

Data collected from interviews, observations, and other resource documents (p. 30) serves as the basis to formulate theories that can interpret or predict expected outcomes. One-onone interviews can uncover biases, historical beliefs, preferences, and specific examples leading to formed opinions. The interviews can also reveal possible actions that can influence change. The use of personal observations can inform the research, showing acceptance, avoidance, disagreement or displeasure with questions or proposed solutions. Resource documents can add

an additional layer of insight; for example, a State of Maine transportation assessment in other regions of the state provided crucial validation during the formation of the theory.

The grounded theory research requires an immersion of myself into the requirements of the data analysis as the interviews are performed and data is collected. My local knowledge and familiarity with the subject will encourage an inherent sensitivity and appropriate response to each interviewee, enabling the consideration of appropriate follow-up questions as the interview progresses and the interpretation of data collected. This strategy enables an ongoing reframing of specific questions as each interview progressed (Merriam, 2014, p. 30).

## Setting

The grounded theory study participants were based in Eastern Maine, the location of a TMA that is managed through a regional economic development organization, Eastern Maine Development Corporation. The TMA was formally launched on February 11, 2014 through a public event in Bangor, Maine. Since that date, activity has focused on establishing the TMA and promoting services in Hancock, Penobscot, Piscataquis, Washington, and Waldo counties affecting 133 communities.

One focus of this research study was to query potential rural rideshare board users to determine their awareness of and willingness to use the TMA. A total of thirteen (13) persons representative of the population of focus were interviewed, for example the elderly, disabled, low income, student, and commuting employee; a minimum of two study participants were recruited from each demographic category. A series of interview questions were posed to each rural resident in an effort to ascertain whether the rideshare board is a valid solution to gaps in transportation services. Study participants were identified via conversations with town managers, healthcare providers, senior groups, religious leaders and groups, school

administrative officials and educators, social groups, and other service providers in order to select interviewees that were representative of the targeted population. A clear description of the purpose of this study with the end goal of identifying reliable, stress-free, and safe transportation for underserved populations served as a motivator for cooperation by community partners. This value-added aspect for their clients enhanced their active engagement in identifying study participants.

The application of a grounded theory qualitative research model was used. The grounded theory research design is described by Creswell (2013) as a qualitative research approach that benefits from the input of a cohort of participants or interviewees (p. 83). This approach generates the "grounded" data from the study participants who directly experience the issue of study, and can help explain what happens in the real world while potentially providing the framework for additional study (p. 83). The qualitative input can lead to a definitive description of a process, action or interaction needed to reach the desired conclusions; in this case, an identification of the perceptions and biases against shared or volunteer ride programs and how to reduce barriers to reduce the transportation gaps in Eastern Maine.

The purpose of this qualitative grounded theory study was to explore the perceptions of selected rural residents of Eastern Maine regarding their need for transportation and their willingness to utilize a TMA model to meet their needs and to generate a general explanation (theory) of the process for choosing transportation resources. This research study emanated from preliminary research performed in a four-county region of Eastern Maine over a three-year period (TIGER2 feasibility study, Figure 1). Eastern Maine Development Corporation (EMDC) in partnership with multiple community agencies worked collaboratively to investigate transportation barriers in rural Maine. This feasibility planning study called "Linking the Rural

Regions of Four Counties in Maine to Enhance Transportation Opportunities and Improve Quality of Life," was a federal Department of Transportation, Federal Transit Administrationfunded project looking to identify solutions to close the gaps and provide options for those looking to connect to employment, services, education, socialization, and more. Figure 2 provides a map of the region impacted by the feasibility study.



Figure 2

The project research included a multi-question survey that was circulated to a broad population (approximately 6,500 households in the four-county region of Hancock, Penobscot, Piscataquis, Washington, and Waldo counties) asking questions about transportation gaps and needs; more than 1,800 individual surveys were completed and returned (approximately 27%). Focus groups and one-on-one interviews were also conducted to provide specific examples and more detailed descriptions of the findings presented in the quantitative data. An engaged and committed Advisory Group, in collaboration with the University of Maine Margaret Chase Smith Policy Center, evaluated the data and began to review options that have been employed nationally to determine their adaptability to rural settings. As a result, the project identified an array of 13 potential options spanning a diverse spectrum of solutions, from ridesharing to expanding public transit routes to creating a formal TMA formed to promote current services and identify new options (2013 TIGER2 Report).

The final report offered multiple solutions, any of which were felt to have the potential to begin closing the transportation gaps. Some of those solutions need to be implemented at the community level, some require an influx of new funding, and some could be performed by an identified entity capable of assuming a leadership role. Based on these parameters, the Advisory Group decided to move forward with the formation of a TMA as a viable approach to systemically oversee transportation activities. It should be noted that the TMA would incorporate many of the other recommendations noted.

The new rural TMA was launched in 2014. During the initial year of operation, attention was paid to adequately market the TMA (now called the Eastern Maine Transportation Management Association or EMTMA) to educate community members about its value and how to access the new resource. At the one-year mark, the interviews were conducted.

#### **Participants**

Care was taken to select 13 representative participants from the demographic populations that were identified through the initial project (TIGER2 feasibility study). The survey tool and personal communications depicted areas within the region as having a high number of clients lacking transportation; this focus allowed access to a subset of individuals needing rides and

therefore positioned to experience higher benefit from the TMA services. Study participants were drawn from communities that range in population from a few hundred to no more than 1,500 residents.

In order to identify individuals representative of the population groups of interest (i.e., seniors, disabled, students, commuters, and low-income persons), a variety of stakeholders were contacted to request recommendations for clients to contact for interviews. These stakeholders included service providers, town officials, healthcare agencies, social and senior groups, educators, and secular leaders. The queries assured that each of the targeted population groups were represented in the interview phase of the research.

# **Qualitative Data**

A series of interview questions were developed to support the collection of comprehensive qualitative data. The questions posed are provided:

1. Do you have a working vehicle?

2. Do you drive yourself?

3. Where do you currently live? Did you need to move to a larger community to be closer to services?

4. Have you been hampered by the lack of transportation?

5. Have you experienced difficulty getting a ride to a healthcare appointment in the last 60 days?

6. Have you missed a healthcare appointment because you couldn't find a ride in the last 60 days?

7. Who has been available to give you rides in the past?

8. Do you feel that the lack of public or other transportation options in Eastern Maine is an important issue that needs to be addressed?

9. Are you familiar with the Eastern Maine TMA?

10. Are you familiar with community rideshare boards?

11. Would you consider using a rideshare board to access a ride in the future? If no, why?

12. What would your primary concerns be for using a community rideshare board?

13. Do you have access to the internet?

14. Have you visited the GettingThereMaine.com web site?

15. Are you familiar with agencies and groups in your community that offer rides to those in need?

A total of 13 interviews were scheduled to be performed at central public locations within each community to make it easier for participants to participate. All interviews were taped and transcribed by an independent transcriber to collect complete and accurate responses to each question.

Enhancing the literature review was a second level of interviews performed with healthcare-providers, healthcare offices and community centers, other service providers, and transportation professionals to learn about agency preferences, perceptions, perceptions of barriers, and acceptance of the proposed TMA model. Interview questions were prepared in advance, and each interview was taped and transcribed to assure the accurate description of each subject's responses.

Although a detailed analysis of the methods required to integrate TMAs into state-level short- and long-range planning efforts was not performed, a series of interviews were executed to

assess biases and perceptions to the potential solutions provided through application of rural TMAs. Interviews were planned through direct telephone communications with the Maine Department of Transportation Commissioner David Bernhardt, Sue Moreau (Transit Policy Specialist, MDOT), Penobscot Community Health Care (federally qualified health center) Director of Patient Services, City of Bangor Health & Community Services Director, Healthy Maine Partnerships Regional Coordinator, Eastern Maine Agency on Aging Executive Director, and Eastern Maine Healthcare Systems Vice President of Business Development.

#### Analysis

An analysis of the interview data was performed to identify the preponderance of observations and values. The analysis of the qualitative data was representative of the information collected and transcribed, and included a subsequent assessment of common threads and conclusions. Bloomberg (2012, p. 137) notes that the analysis of grounded theory data requires that categories for comparison are formed (called a process of open coding) and are analyzed in relation to the projected theory, following which the stories (interviews) are weaved to build an interconnection between the interviews. Each of the thirteen interviews was labeled as Study Participant 1, Study Participant 2, etc. to assure anonymity. Within each interview, categories were identified to isolate thoughts and comments related to primary topics of consideration including vehicle ownership, factors contributing to the lack of transportation, transportation resources, missed healthcare appointments, familiarity with rideshare boards, familiarity with TMAs, and biases and preferences for rideshare options. A further segregation of the data was performed to identify key responses (performed by building a spreadsheet and populating each question posed with responses and trends).

Complementing the data collection process, the application of the constant comparative method of data analysis was employed. This process consists of the comparison of segments of data with the other segments of data collected to determine parallels and differences. As a result of these comparisons, categories of evaluation were formed to facilitate the identification of patterns in the data. The observations and conclusions support the development of a new theory that addresses questions about process, in this case the changing dynamics of rural transportation opportunities (Creswell, 2013, p. 86).

#### **Participant Rights/Confidentiality**

All consideration was made to assure the confidentiality of study participants. If provided, names were omitted from the data reporting summaries. Interview study participants were transcribed as Participant #1, Participant #2, and so forth. Each respondent was informed of the purpose of the research, value of their participation, and what will happen with the data collected. Participants were informed that they could stop the interview at any time, and could withdraw their participation if they no longer wish to participate.

## **Potential Limitations of Study and Conflict of Interest**

Extreme care was taken to limit researcher biases and address perceived conflicts of interest. The interviews were performed using an extensive script; the script was provided to independent reviewers for input prior to the interviews. Any perception of bias or preference might be evident in the pattern of questions, but that data itself was objectively transcribed. Although no problems were anticipated in recruiting professionals for one-on-one interviews, there was the risk that residents would be hesitant to participate. This risk was felt to be low as residents were queried to participate in the original planning study to assess transportation gaps and potential solutions, and no lack of volunteers was noted.

## **Pilot Study**

The interview questions were prepared and tested in a sample group to assure that the data collected addressed the questions of concern; this sample group was planned for 5 individuals, and included clients and staff at Eastern Maine Development Corporation as well as other community members. Care was taken to assure that pilot study participants were not located at one of the communities or agencies selected for this research study.

A total of five participants were recruited for the pilot study. Of the five, three were Eastern Maine Development Corporation employees and two were actual study participants recruited by the referring agencies. The outcomes of the interviews for the two non-employees were included in the final evaluation. The pilot interviews were conducted similar to those planned for the research study; interviews were held in a private room, a tape recorder was placed in clear view of the participant when the conversation was recorded, the questions were posed following the script, and the descriptions for GettingThereMaine and the rideshare board were identical.

Through the course of the pilot study, it became clear that a description of the EMTMA and rideshare board would be an important piece to inform the interviewee and prepare them for the interview questions. These descriptions were fine-tuned with input from the initial pilot participants, and were finalized for presentation to the final two pilot participants. An additional clarifying question was added to the script to ask about relocation from the rural home to a service hub.

## **CHAPTER 4**

## **RESEARCH FINDINGS**

The purpose of this grounded theory qualitative study was to explore the perceptions of selected rural residents of Eastern Maine regarding their need for transportation, and their willingness to utilize a Transportation Management Association (TMA) model and alternative transportation options to get where they need to go. This study centered on rural residents' perceptions of a TMA, introducing the purpose of rideshare boards, and gaining insight on whether usage of this resource is a feasible solution for people living in Eastern Maine.

Because of the extreme rural geography of Eastern Maine, alternative solutions to the traditional public transit model are needed to bridge the transportation gap and assist those needing access to rides to a wide range of destinations including healthcare, social services, employment, school, shopping, childcare, and socialization activities. The gap in services is amplified by the need to travel to access services at centrally located "hubs" such as Bangor or Ellsworth, the socio-economic characteristics of rural Maine residents (aging, low-income), and the shift of employment opportunities from rural communities to larger cities particularly in the face of numerous paper mill and manufacturing closures in the last 5-10 years.

This study addressed four research questions: 1. What were the perceptions of the study participants in rural Eastern Maine about the alternatives presented through the TMA including the rideshare board and volunteer systems? 2. Do study participants in rural Eastern Maine needing transportation consider TMA's a realistic solution for meeting their transportation demands? 3. Do study participants in rural Eastern Maine perceive that TMA options will increase their ability to keep medical appointments? 4. Do the

participants indicate that they will be likely to access TMA options to keep medical appointments?

Central to this study was the identification of factors that limit access to services in an effort to understand the barriers, availability of supportive structures/services, and analysis of ingrained perceptions that could negatively or positively influence usage of the TMA model. The six factors identified to guide the interviews and frame subsequent analysis included:

1. The identification of socioeconomic status, and transportation characteristics and impacts;

2. The elements impacting rural transportation including identified barriers to mobility and characteristics that influence these barriers;

3. A review of multimodal options beyond public transportation;

4. The formation of TMAs that are driven by community partnerships and collaboration, and the review of critical components that comprise the successful TMA;

5. A review of recommendations from the literature and the critical components of interest;

6. A review of psychological factors that can influence behavioral change.

The research focused on study participants who live in rural communities in a fourcounty region of Eastern Maine including Hancock, Penobscot, Piscataquis, and Waldo counties. Thirteen persons were recruited and interviewed by asking a series of 14 questions. Care was made to not interject any interviewer biases or perceptions during the interviews. Clarification as to the definition of a TMA, rideshare boards, and the GettingthereMaine.com web site was provided in an effort to inform the study participant only. Each interview was recorded and

transcribed for accuracy and to attain a true representation of interviewee responses to each question.

This chapter summarizes the study participant responses and reviews the results of the data collection, with particular attention given to how the responses correlate and answer the research questions. Key themes will be identified, and chapter 5 will include the impacts of the data and recommendations.

#### **Study Participant Interviews**

The interviews were performed to gain insight about study participants' knowledge about current transportation options through the TMA and the rideshare board posted on the GettingThereMaine.com website. The interviews also served to query study participants about their familiarity with other options available in the region to get those in need to their destinations.

The original research design included scheduled interviews in a neutral location where the study participant would feel comfortable and safe. To some extent that protocol was useful: 7 interviews were performed in open and public locations such as the City of Bangor Health & Community Services offices, Beal College, and the Hammond Street Senior Center, all located in Bangor, Maine, a hub service center in Eastern Maine. However, six of the interviews were performed at the residences of the study participants per their request as this proved to be the most convenient location since none had readily available transportation.

The study was designed to interview participants who currently live in small remote and rural communities in Eastern Maine. This proved challenging since many residents most in need of transportation had already relocated to the Bangor community. As such, of the thirteen (13) study participants interviewed, eight (8) resided in Bangor and the remaining five (5) were from

surrounding rural communities. Of importance, two of the eight Bangor residents were native to the service center; the remaining six had relocated to Bangor within the last 15 years to be closer to services, family, school or work. The geographic relocations to the larger town(s) were all directly related to the lack of transportation.

Another variation from the proposed research was the plan to recruit patients directly from the Bangor-based federally qualified health center Penobscot Community Health Care (PCHC). As a health care center dedicated to providing healthcare services to low-income and underserved clientele, their participation was viewed as critical in order to assess the difficulties associated with keeping medical appointments. Unfortunately, PCHC did not participate as originally planned and they did not refer participants to the study. Because of PCHC's lack of participation, I adjusted my discussions with local service providers to emphasize the goal of talking about missed healthcare appointments. This proved effective as many providers serve as case managers and have a deep understanding of the barriers and needs of their individual clients. I was, therefore, able to gather information related to missed healthcare appointments for the research study as originally planned.

An assessment of the impact of a lack of transportation on missed medical appointments was performed as four of the 13 study participants were challenged by on-going, chronic health issues including one individual who has been hampered with life-long polio and has been wheelchair bound since an early age.

In order to identify individuals with the selected characteristics, contact was made with professional colleagues who assisted in the process. Service organizations that provided study participant referrals included:

• The *City of Bangor Health & Community Services*, the regional social services office that provides traditional welfare services to low-income individuals including Shelter Plus Care (Section 8 housing and counseling services), the Women, Infants and Children program (WIC), and social services counseling to residents of 21 rural communities surrounding Bangor Maine.

• *Penquis*, a Bangor-based community action program that provides a wide range of services including LiHEAP fuel assistance, housing, daycare and early childhood education services, and transportation services (LYNX) for Medicaid and Medicare recipients.

• *Beal College*, a private, small technical school that provides training and education for in-demand careers for about 500 students annually. The school provides accredited one-year and two-year programs for careers including health information management, accounting, medical assisting, criminal justice/law enforcement, and substance abuse counseling to name a few.

• *Eastern Area Agency on Aging (EAAA)* – an agency providing a range of services to seniors, adults with disabilities and caregivers for over 40 years. Services include caregiver case management, dementia programs, aging and disability resources, nutrition, wellness programs such as falls risk assessment, balance, and Tai Chi, Meals on Wheels, and a variety of programs for socialization.

• *Hammond Street Senior Center (HSSC)*, a Bangor-based meeting place for active, independent seniors including those living in Bangor as well as 21 surrounding rural communities. HSSC provides programs that are unduplicated in the region and focuses on enabling seniors to age in place. A sampling of activities includes the Senior University with classes such as physical fitness and balance, arts and crafts, computers, dance, and continued learning, field trips, social events, and access to affordable home services.

Despite these challenges and modifications to the original research study, I was able to conduct the research with representation from the primary population groups of focus:

- 1. Low-income participants 8
- 2. Physically disabled -3
- 3. Students -2
- 4. Commuters -1
- **5.** Senior Citizens 4

# **Participant Profiles**

Thirteen study participants were interviewed for this research study as the result of referrals from local service providers. Brief participant profiles are provided in Table 1. Each participant was assigned a number in order to assure confidentiality and anonymity.

At the start of this research project, I contacted local and regional social and healthcare service leadership to introduce my research plan and methodology. The goal was to familiarize my colleagues and to engage their assistance with the referral of rural residents. Follow-up was either by telephone calls or visits to each service provider to share an outline of my research to fully inform them of the process and goals. This proved to be a useful approach and resulted in the referral of an adequate number of participants; the service providers were able to assist with the recruitment process by explaining their participation as well as introducing me to the participant. Although the initial referral was made by the local service provider, direct contact was made with the study participant to schedule a day, time and location most convenient to them. In all cases, I traveled to the participant to assure convenience as well as actual participation in the study.

# Table 1

# Participant Profiles

Study Participant	Gender	Age	Population Group	Transportation Challenges
#1	Male	40	Unemployed, Low-income	No vehicle, no license
#2	Female	46	Unemployed, Disabled	Unable to drive
#3	Female	20	Unemployed, Low Income	No vehicle
#4	Female	48	Unemployed, Low Income, Disabled	No vehicle
#5	Male	74	Retired, Senior Citizen	No vehicle
#6	Female	71	Retired, Senior Citizen	No vehicle
#7	Female	92	Retired, Senior Citizen	Unable to drive, no vehicle
#8	Male	69	Low-income, Disabled	No vehicle
#9	Female	70	Retired, Senior Citizen	No vehicle, unable to drive
#10	Female	45	Student, Low-Income	No vehicle
#11	Female	25	Student, Low-Income	No vehicle
#12	Female	31	Low-Income	No vehicle
#13	Female	72	Low-income, Retired, Senior Citizen, Disabled	Unable to drive

# **Analysis Methodology**

Each interview was prefaced by a brief overview of the purpose of the research, why that individual was selected to participate, an explanation of voluntary participation, and steps that

would be taken so complete confidentiality would be maintained. Audiences that might view the outcomes were discussed. A tape recorder was placed directly between myself and the study participant to assure that their words were recorded as stated and their comments would be shared as they intended. Interviews ranged in length from 15 minutes to 45 minutes. As part of a pilot phase, the initial two interviews were performed and then evaluated and coded to assure that the questions asked led to relevant responses needed to support the development of conclusions. The questions as asked were found to be inclusive and clear, and the responses would support conclusions to answer the research questions posed.

Immediately following each interview, the tape was transcribed by an independent transcriber contracted to provide this service. The completed transcription was read and compared to the audiotaped version to assure that the interview was accurately transcribed. During this review, care was taken to notice my preferences and bias and to eliminate redundancies. Additionally, each transcribed interview was coded to reveal trends and shared responses that would help with the characterization of the analysis. At the completion of the transcription and coding of the thirteenth interview, the interview responses and codes were reviewed and tabulated to identify trends and patterns similar across all study populations.

Segmenting responses to the research questions allowed for the coding of the interview data. Creswell (2013) suggests that separating the data into small categories of information enhances the analysis process. Saldana (2013) also notes that coding can be a method of discovery and can serve to organize and classify the data into emerging categories for further analysis.

Using this approach, the responses to interview questions were reviewed with key words and phrases highlighted. For example, responses to Question #12 (What would your primary

concerns be for using a community rideshare board?) resulted in the following phrases being highlighted as common threads of thought: Safety; fear; stranger; reliable for returning home; cost for ride; references; background check; alone; female rider; and, safe car. Each question was similarly reviewed and coded to discover emerging themes.

# **Common Themes**

As the interviews were analyzed, common themes threaded through the conversations. Regardless of their socio- or economic circumstances, these themes were consistent among the interviewees. Briefly, these included:

Table 2 Common Themes

Theme	Observation	Impacted Participants
1	Participants had already relocated to urban hubs	#1, #2, #8, #9, #10, #12
	(such as they are in Eastern Maine) or service centers due to	
	a lack of transportation and ready access to jobs, education,	
	services, shopping, social integration and cultural inclusion.	
2	A fear of the unknown and personal safety are	All Participants
	prohibitive factors to seeking or accepting rides.	
3	The lack of awareness of available transportation	All Participants
	resources is a barrier to using the TMAs rideshare board.	
4	Creation of an environment to promote usage is an	All Participants
	important approach to encourage increased use of the TMAs	
	rideshare board.	
5	The lack of reliable transportation has resulted in	#2, #3, #4, #6, #7, #10,
	missed healthcare appointments.	#11

6	Substance abuse issues or situations resulting from	#1, #2, #4, #8
	substance abuse can be prohibitive to owning a vehicle and	
	driving, factors contributing to the lack of transportation.	

#### Theme #1: Relocation to Service Centers was Common

One finding from the study was that, for many of those interviewed, the decision to relocate from a remote and rural community to be close to a wide array of services including healthcare, jobs and job search assistance, housing, education, and counseling had already been made. Because of the need for proximity to structured "help" and the difficulties arising from a lack of transportation, many rural residents in the study had already made the difficult decision to leave their historical homes and communities to be closer to needed services. Of the 13 study participants, six had already relocated to Bangor (6 of 13). One study participant noted that his relocation was not by choice: "I've always lived in the country, but I've always had trouble… mainly all my trouble's been driving. I don't have a driver's license, so I am forced to live in town just to be close to anything or anything with work" (Study Participant #1). This individual also pointed out that, even relocating into a city with public transportation, his difficulties associated with rides to work have not been resolved as many jobs that he has pursued require evening shifts when public transportation is not available.

Another participant relocated to Bangor in order to be closer to family (Study Participant #8). His sister has become his mode of transportation for shopping, health appointments, social services, and other needs. It became too difficult for his sister to travel the miles from Bangor to his rural community to pick him up, bring him to Bangor for an appointment, and then transport him back home. Because of the lack of public transportation, his only other option was taxi

service which proved too cost-prohibitive for him to use as a reliable and affordable backup. Therefore, he relocated to Bangor for convenience, both for him and his sister. His relocation hasn't totally eliminated his need for taxi service, and he is forced to pay out of pocket for cab fares when his sister is unable to transport him to destinations.

Yet another participant shared that she recently moved to Bangor due to health issues prohibiting her from driving a personal car (Study Participant #2). As she had been commuting an average of 22 miles per trip to drive from her home to Bangor for social services/counseling and healthcare, she decided that it made sense for her to uproot and relocate to reduce some of the stress related to mobility. The public transportation system in Bangor allows her to get to her multiple appointments with some ease without worrying about rides to and from different locations across town.

These factors limiting transportation options were common to other study participants who echoed the need to move to reduce the stress of finding rides, saving money previously used for expensive taxi fares, and assuring that appointments could be kept.

Of the study participants interviewed, two had lived their entire lives in Bangor (Study Participants #4 and #10). Although they were not driving from far distances to acquire services, school, or employment, they experienced a high level of inconvenience by not owning a personal vehicle and having to rely on public transportation with limited service hours and/or taxi service. Both of these solutions proved to be challenging both in terms of cost and convenience.

As a result of the lack of transportation, relocation from rural communities to larger service hub communities was viewed as the best option for some participants to resolve the lack the transportation and enhance access to needed services.

#### Theme #2: Fear of the Unknown and Personal Safety were Prohibitive Factors

Although the fears associated with traveling in a car with a stranger were more pronounced with women and seniors, every study participant introduced to the concept of sharing rides indicated they had concerns with the basic principles of a rideshare board.

Study participants were skeptical when first introduced to the concept of rideshare boards as a solution to closing the transportation gap. Much of this reluctance was due to a fear of the unknown – getting into a vehicle with someone they didn't know. One participant noted that she would be hesitant to use the rideshare board "because I'm a woman who lives alone and I'd want them screened." This individual also noted that while needing transportation to work, "even people with a criminal past can work and own a car" (Study Participant #6).

Another participant shared "Honestly, I probably wouldn't, I'm stubborn, it would be a last resort if I did just because of the thought of I don't know who I'm riding with" (Study Participant #1). Similar concerns were voiced by all study participants.

The option of screening the drivers was mentioned by four study participants, either by suggesting that agencies use background checks or a similar vetting process. There was one concern related to social anxiety, while another voiced fears because of a past with domestic violence. She noted that assuaging those fears would be difficult because, as she stated, "It's one of those things that never completely goes away, you just try not to be hyper-vigilant about it. You just kind go...I'm going to use my common sense and if something doesn't seem right, I am going to listen to that instinct and just say you know what, thanks anyway, I'm good" (Study Participant #2).

Only one participant noted that there would be a concern about the safety of the vehicle itself as opposed to the driver of the vehicle (Study Participant #12).

Fears and doubts related to personal safety were primary issues presented by all study participants.

## Theme #3: Lack of Awareness of Available Resources is a Barrier

To a person, none of those interviewed were aware that a local rideshare board was available to them as an option to close the gap for rides. Although two participants thought they might have heard about the EMTMA and/or GettingThereMaine.com web site, none were familiar with the transportation services provided or with the concept of rideshare boards.

There were two study participants who had heard of or used the Lynx program, which is a free ride service for individuals in Eastern Maine who are enrolled in the state-supported Medicaid program called MaineCare or the Medicare program. Unlike the GettingThereMaine rideshare board, drivers are recruited, screened, extensively trained, and paid by Lynx program resources. Of the 13 study participants, two had accessed transportation through the Lynx services. The Lynx service is not without problems according to Study Participant #2. Although this individual did not directly experience the problem, a friend had shared that her father had scheduled a ride for blood work and when the driver never appeared for the ride back home, a call was made and the response was that Lynx was not aware that a ride home was needed. This resulted in the friend's father needing to make alternative arrangements to get home.

The executive director of the Eastern Area Agency on Aging (Executive Director, Eastern Area Agency on Aging) pointed out concerns with reliability with the Lynx system that have been shared with her in the past. Senior clients have consistently reported that rides have been cancelled by the agency because of low ridership that day.

Because of the lack of other options available to close the transportation gap, all study participants have continued to look at other alternatives including cab service, rides with family or

friends, or more often rearranging of schedules to match ride availability. The prohibitive cost of cab service and the complications associated with rearranging appointments were viewed as poor alternatives. Turning to relatives and friends was also difficult; interviewees were concerned that they would be viewed as burdens.

The lack of familiarity of options in Eastern Maine contributed to the difficulty in finding reliable solutions for rides, indicating that improved outreach and education is needed to fully inform the public of available options.

#### Theme #4: Creation of an Environment to Promote Usage is Important

Another common theme that surfaced during the study interviews was the need to build a positive atmosphere that is supportive of the alternative transportation methodologies. It should be noted that only one of the study participants addressed the issue of having friends and neighbors aware that they were in need of help. Specifically, this individual (Study Participant #9) was concerned others would think poorly of them if they knew they were dependent on others to get to their desired destinations. In spite of the fact that only one study participant specifically pointed out this issue, care and service providers in this region consistently discuss the barrier of getting services to impacted persons because of the negative stigma associated with having needs. The perception of most is that participants would rather go without services, including transportation, than let others become aware of their problems. The executive director at the regional agency on aging (Eastern Area Agency on Aging) shared that many of her clients are willing to go without healthcare services, social services, or healthy food choices rather than give friends and family a sense that they need help (interview). She also noted that even those who are approached and offered help refuse the offer because they don't want to seem helpless or in need.

Study participants were uniform in stating that an approach that promotes the benefits of using the rideshare board and EMTMA services, including testimonials that others have used the services, would be very influential. Approaches to change attitudes specifically those with concerns about safety and reliability of the ride are necessary if usage is to increase. When speaking with the interviewees, all agreed that sharing rides was a good solution to reduce car emissions and the overall carbon footprint in the environment. Describing rideshare boards as beneficial for the environment was viewed as a more palatable reason to explain their use as opposed to the financial or assistance needs of the user.

# Theme #5: Lack of Reliable Transportation has Caused Missed Healthcare Appointments

One of the questions posed by this research study was whether the lack of transportation contributed to individuals missing healthcare appointments. Of the 13 individuals interviewed, seven had experienced missed appointments due to the lack of a ride. Study Participant #2 shared that, although the Lynx system offered by the Bangor-based community action program is helpful, "Nothing is perfect and sometimes there is a breakdown of communication, maybe you got your ride set up, then your ride doesn't get there on time or does a no-show."

Another participant shared that she has psychiatric appointments that are hard to schedule, and it often takes months of waiting to finally get an appointment only to have to cancel or do a no-show because she couldn't find a ride or didn't have gas money (Study Participant #3).

The risk of multiple missed or last minute rescheduling of appointments is critical as some medical providers can take action and dismiss the person as a patient. An alternative action could be a delay in renewing prescriptions that results in a lapse of medical treatment. One study
participant said that although a healthcare appointment hadn't been missed, the fear of not making an appointment is an additional stressor to a life that is already worrisome (Study Participant #6).

Missing a healthcare appointment can have many negative impacts for the rural resident including an interruption in the quality and timing of their care, diminished success in the management of chronic diseases due to lapses in therapy, missed medication doses due to expired prescriptions, and an increased risk of dismissal of the provider due to multiple no shows.

# Theme #6: Substance Abuse Issues can be a Prohibitive Factor

Prevalent with many study participants is that past and/or present problems with substance abuse have had a negative impact on their transportation independence and mobility. One 20-yearold participant noted that, although she was trying to get on her feet again and do better including looking at going to the regional community college and looking for a job, her past substance abuse has led to her eviction from one location and she is now couch-surfing with friends (Study Participant #3). Her recovery is hampered by her difficulty getting to counseling, including psychiatric intervention.

Another study participant no longer had a valid driver's license as it was suspended due to drunk driving infractions and driving under the influence of other substances (Study Participant #1). As a hub center for social services and counseling in this region, residents often need to travel to Bangor or the adjacent community of Brewer to get preventive and supportive intervention counseling. The gap in transportation services is a significant barrier preventing access to services needed to assist these individuals and support their re-integration as a productive member of the community.

Six common findings from the data include: the relocation of interviewees to service hubs, a fear of the unknown and need for assurance of personal safety, a lack of awareness of available

resources, the need for an accepting environment that influences behavior, a lack of rides resulting in missed healthcare appointments, and the impact of substance abuse were prevalent throughout the course of the research study. Each study participant willingly shared their past and current barriers as well as the lack of viable options in the hopes that their participation in the study would result in more alternatives to close the transportation gaps they face each day. The representative population of Eastern Maine's rural residents confirmed that the identification of viable options must address or at least consider these themes when building an effective TMA.

## **CHAPTER 5**

# CONCLUSIONS

One study participant made an astute observation during the course of her interview. Due to her physical handicap, she took the time to describe the lengths that she must take to assure that she has rides to medical appointments, social service appointments, the grocery store, and even social interaction. She equated planning for transportation needs similar to planning for one's own funeral: No one wants to think about the day when having to plan for transportation or your own funeral comes, but at some point it's unavoidable. This difficult decision signals that that the independence of driving oneself where he/she needs to go is gone. Having that independence is often unrealistic for many residents in Eastern Maine. Also unrealistic is the idea that the State will be able to address these types of transportation challenges through an expansion of funding for increased public transportation.

The preferred mode of transportation would be ownership of a personal vehicle with resources to cover all the expenses that are required with vehicle ownership and the ability to drive oneself to points of destination: car insurance; funds for gasoline; and, funds for car maintenance. The reality is that, for many, these barriers to personal transportation are insurmountable. Although all participants acknowledged that owning a vehicle and driving independently was not likely, it was hard to face the alternative of asking for help and relinquishing that independence once and for all.

The goal of this research study was to determine whether the Transportation Management Association (TMA) platform could serve as the conduit to connect people needing rides to those able to provide assistance. The conclusions presented reflect the interviewee responses and trends

or commonalities, and magnify the overwhelming need for new solutions in rural communities to narrow the transportation gap in Maine.

Four research questions were developed to guide this research study:

<u>Research Question 1</u>: What were the perceptions of the study participants in rural Eastern Maine about the alternatives presented through the Transportation Management Association (TMA) including the rideshare board and volunteer systems?

This question would assess whether study participants were aware of the existence of rideshare boards, and in particular were informed about the Eastern Maine TMA rideshare board or the GettingThereMaine.com web site. If there was a lack of awareness, would participants be open to the concept and trigger their personal use?

<u>Research Question 2</u>: Do study participants in rural Eastern Maine needing transportation consider TMA's a realistic solution for meeting their transportation demands?

This question would discern whether acceptance, skepticism, or refusal to consider this option by study participants would be observed, and would identify factors influencing their acceptance or dismissal of this alternative as an option for future use.

<u>Research Question 3</u>: Do study participants in rural Eastern Maine perceive that TMA options will increase their ability to keep medical appointments?

If a consensus of study participants' voice their experiences with missed healthcare appointments, would TMA's be viewed as an acceptable solution to quell their concerns and increase their ability to keep appointments?

<u>Research Question 4</u>: Do the participants indicate that they will be likely to access TMA options to keep medical appointments?

Regardless of their initial concerns, this question would identify acceptance of the rideshare board and the TMA as an option to address their future needs. Should the response be negative, reasons for their lack of acceptance would be discussed.

Participants stated several areas of concern that need to be addressed for a TMA to successfully connect those needing a ride to resources that can fill that need. An increase in the visibility of resources, both those provided through a TMA and resources available through other agencies is needed. The fears and concerns about personal safety are at the forefront in most residents' minds, and a realistic and compassionate response is needed to convince users that their concerns have been addressed. Challenges inherent to technology, including on-line access to services, are a real barrier for all ages and incomes of users. Providing digital literacy and adequate support to train users on the methodology is needed.

Missed healthcare appointments and the resulting impacts (e.g., loss of the medical home, interruption of healthcare interventions, and suboptimal chronic disease management) is a serious problem for those without reliable transportation. For those with physical handicaps, the additional transportation needs can be prohibitive to accessing the needed care. Mental health and substance abuse services can be particularly challenging as the prescribed intervention often consists of multiple appointments per week for optimal management.

Implementing measures to begin adjusting behaviors and perceptions is a necessary component to building confidence in the TMA. Often these measures include adequate outreach and education to increase awareness and convince potential users of the benefits to accessing transportation via alternative modes. The methodology employed to accomplish the outreach and education should be based on building trust and confidence, and should be multi-modal in nature (i.e., reading materials such as pamphlets and brochures, media outlets/stories, and public service

announcements). While sharing the benefits of using a rideshare board is important, building confidence and a sense of security will likely drive the decision to use or not use this option.

Through the course of this research, several salient points were noted that characterize the nature of transportation in Eastern Maine, and what transportation in the future will look like for people living in rural Eastern Maine. Making the connections necessary to get people to their destinations will be supported by a change in attitude, a change in behavior, and an identification of alternate resources to reduce many if not all of the barriers noted. One solution is the implementation of a TMA or a rideshare board; developing a plan that can serve to ultimately modify behaviors, particularly behaviors that are born out of historical biases and preferences will be required.

#### **Increased Visibility of TMA**

Although the GettingThereMaine web site and rideshare board have been active for over a year, the response from study participants indicates that the message is not reaching the appropriate audience. Of interest, those organizations referring clients for participation in this study have all been actively engaged in the EMTMA and rideshare effort; yet a disconnect between the agency and clients is occurring and information regarding the availability of transportation options is not reaching those in need.

Following the conclusion of the interviews, the referring agencies were contacted to inquire if additional measures were needed to get the word out. Tools such as brochures and posters were viewed as positive, but were not serving to adequately inform the public. Their recommendations included: finding a strategic location for viewing of materials; linking the social service agency web site to the TMA GettingThereMaine web site; scheduling of on-site meetings to share information about the TMA/rideshare board including availability to answer questions and

demonstrations for using the rideshare board on-line; and, use of marketing tools such as GettingThereMaine buttons to hand out to clients during agency visits.

It was also recommended that the TMA staff increase their participation at local and regional venues such as healthcare fairs, job fairs, and special community events such as the local Chamber's annual Business Expo, the Senior Expo, and other socialization opportunities.

Increased marketing with local media outlets was encouraged including preparing information for advertisements and articles in the weekly free newspaper, and public service announcements on radio and television. These are all venues that are easily accessible and free to the consumer.

#### **Increased Confidence With Safety Issues**

The biggest barrier to overcome is one of safety. Although prevalent with women, elderly men have shared the concern that they may not be safe when entering a vehicle with a stranger. Those fears include physical safety but also expand into concerns with abandonment and financial scams. There is a tangible fear about getting dropped at the destination and not getting picked up to return home. In rural communities where taxi cabs are not always available, the idea of being stranded without options to get home is a real concern. Similar to abandonment, there was concern about accepting the ride and then being forced to pay when the ride is concluded, particularly in order to assure that the driver will return for the ride home. Although the premise of the rideshare board is often based on free transportation, eliminating that fear is important to increase confidence with this transportation option.

Overcoming those fears will be challenging. One example employed to reduce fears is demonstrated with the recently implemented employer portal site on the EMTMA which connects employees needing rides to those offering rides. As a safety feature, employees are encouraged to

contact supervisors within the organizations to confirm employment and to obtain a character reference. For these riders, getting into the vehicle with a fellow employee adds a layer of safety to the partnership.

For others accessing rides, instructional videos can be used to instruct users on the best methods to connect with rides. As an example, getting a ride with someone from the same or nearby community tends to reduce apprehensions as rural Eastern Maine is considered a small community where everyone knows everyone, either directly or indirectly through family or friends. Encouraging users to ask for references from employers, churches or other social clubs is another way to allay fears.

In discussions with interviewees, I asked their opinion on what would reduce their fears of getting into vehicles with a stranger through the rideshare board. Suggestions included the performance of background checks on drivers and riders if requested, offering a venue to meet prior to the ride, and marketing the safety of rideshare boards through published testimonials. As one female interviewee noted, having an opportunity to meet the driver, look into his/her eyes, shake their hand, and talk for a few moments would increase her confidence in getting into the vehicle on the day of the ride. Another suggested that she would take a friend along for the first ride to increase their sense of well-being.

Concerns with safety can't be ignored and, if effectively addressed, will likely support increased use of the rideshare board.

#### **Technology Challenges**

For seniors and many low-income people, having access to or understanding how to use the GettingThereMaine web site to meet their transportation needs is challenging. Digital connectivity and literacy is a barrier in Eastern Maine (Re-Engineering the Region Final Report, 2015), and

many residents are not connecting, either because they do not have a computer, do not have connectivity infrastructure in their rural community, can't afford the internet service, or don't know how to use the internet.

Eliminating this barrier will be key to increasing access to the GettingThereMaine web site and services. Training needs to be available at the service agencies as well as the local public library. Toward that end, developing a user guide that provides step-by-step instructions will be useful. In addition, teaching the resource providers at service agencies (e.g., case manager, healthcare provider, educator, library staff and others) about using the GettingThereMaine rideshare board, employer portal, and other resources will broaden the expertise beyond the TMA staff.

### **Missed Healthcare Appointments**

Successful management of one's health is predicated by the ability to maintain a care management plan that can be followed. The lack of transportation often minimizes that success because of missed appointments resulting in ineffective management, particularly for those with chronic diseases or those undergoing counseling for substance abuse or mental health issues. For those requiring ongoing intervention (for example, a patient on weekly dialysis), keeping scheduled healthcare appointments can mean the difference between a healthy outcome and future hospitalizations. Partnerships between healthcare providers and TMAs can be one solution to reduce barriers to assure that each patient is receiving the best possible care.

Educating patients will need to occur at the provider's office, with the healthcare professional trained to describe and instruct the patient on how to access the rideshare board for future appointments. In Eastern Maine, efforts are underway to connect case managers with the

TMA to share the outreach tools and provide basis instruction on use of the web site which can be shared with the patient.

# **Behavior Modification**

There has been considerable conversation around the idea that Mainers are independent, don't like to be told what to do, and are reluctant to share their difficulties and need for help with others – even family. As for transportation, the goal is to own a private vehicle and have the independence to come and go as you need. This is unrealistic for many Eastern Mainers, particularly seniors and the disabled who are physically unable to drive and the low-income individuals lacking financial resources to purchase a vehicle and the ancillary costs of gas, car maintenance and car insurance. Moving beyond these self-imposed barriers will involve a reeducation process to convince people that admitting they need a ride is not a sign of weakness or shameful. More effective will be the use of a "neighbor helping neighbor" attitude. For seniors and the disabled, the idea of sharing a ride as a socialization opportunity rather than a burden was appealing to these populations when interviewed.

Increased usage will be influenced by the way the TMA transportation options are introduced and described. Testimonials by "regular" people that show positive outcomes from using the rideshare board will send a positive message – it is not painful, embarrassing, or an admission of one's inability to take care of their own needs.

Another method to break the behavior barrier would be the promotion of environmental benefits. The reduction of the fossil fuel footprint is a positive outcome of ridesharing, and promotes the quality of life enjoyed by Mainers. Stating that use of the rideshare board is helping to preserve clean air and reduce one's reliance on fossil fuels can be a positive message.

### Methodology for Outreach

As noted earlier, implementation of an outreach and marketing plan will be critical to the success of the rural TMA. Using a variety of tools including brochures, posters, waiting room videos, and information cards help to get the message out to those in need, particularly if these materials are posted for viewing in locations accessible to the intended audiences.

Using testimonials from other users can be a viable approach. Sharing one's personal story about using the rideshare board, including a description of their initial fears, learning the process, screening prior to the ride, and outcomes of the ride can begin to build confidence with others that this option could work for them.

Working with the media outlets will be critical. Public service announcements can be produced in partnership with the media and the TMA to build content that is useful and relevant to potential users. Determining airtime is critical to assure that the message is getting to the appropriate audiences when they are most likely to see the announcement.

Linking web sites with community service partners' will also help drive users to the GettingThereMaine web site. While there may be apprehension by some to using the TMA web site, familiarity with the direct service provider web site could influence and drive potential users to visit the transportation site.

## **Concluding Observations**

Through this research study, issues were uncovered which may ultimately drive whether this alternative transportation solution will be successful. Foremost of these was the clear message that information about the Eastern Maine TMA was not reaching the intended audience. There is a real disconnect between what the services providers are sharing and what the potential users are hearing. As noted by Morris (2006), a well-planned and clear marketing plan is needed to increase

awareness and community recognition. Employing a "push versus pull" strategy, also recommended by Morris, will be critical. Rather than waiting for the clients to voice their need, the service providers and TMA staff need to actively share the resources and options available through the TMA. Proactive education will assure that appropriate resources can be effectively accessed when the need arises. The approach will involve more than sharing brochures or putting up posters. Discussing the alternative options with clients as part of the service visit should be routine to assure that each individual is aware of the TMA and the GettingThereMaine web site. This strategy will be reliant on the successful partnership between public and private partners in the communities who collaboratively share the message to close the transportation gaps.

Substance abuse is another issue that can negatively impact the ability to get to desired destinations. Whether through the loss of one's drivers' license or loss of the ability to own a vehicle because of the financial realities of substance abuse, the ability to get where you need to go is severely hampered because of the long-term impacts of drug and alcohol use. This negative impact extends beyond recovery for the user; the need for continued supportive intervention services requiring multiple rides and difficulty securing and keeping a livable wage job is a common theme for this population. Understanding these barriers and increasing awareness and access for this population will be another challenge for the service providers in Eastern Maine.

Also important is the need to interact with healthcare providers to share the study participants concerns and difficulties connecting with reliable rides. Those interviewees are representative of people living in Eastern Maine who are also experiencing transportation barriers, particularly in relation to maintaining their health and well-being. The medical and service communities will need to collaborate to assure that access to healthcare continues as well as combine efforts to open doors for rides in the future. Sharing the education and resources from

many gatekeepers will increase the likelihood that the public is aware of the TMA, GettingThereMaine, and the process to find transportation when needed.

One issue that I suspected may arise but never did was the barrier of finding willing drivers to offer rides. Since all study participants noted that they were unaware of the rideshare board and TMA, I was unable to determine if there was a barrier to actually connecting to someone offering a ride when it was needed. Further investigation is warranted to assess whether a new barrier will surface – as more access the rideshare board to get rides, will the lack of available drivers continue to perpetuate transportation gaps? In anticipation of this, it will be important for service providers and TMA staff to reach out to communities to share the TMA resource and solicit volunteers to offer rides to those in need.

Identifying the issues and overcoming the barriers to access reliable transportation is a challenging but necessary process that will contribute to the formulation of a well-planned approach that begins to get more people to their destinations. The preference for single-driver cars, skewed land use patterns favoring one-driver vehicles, and fears about personal safety contribute to the barriers faced by Eastern Maine residents. The barriers identified directly impact access to education, jobs, healthcare, and socialization, all of which contribute to one's well-being.

Although current alternatives exist, the lack of awareness hampers successful access to services. The utility of the TMA as an option for those seeking rides will be measured by usage. That usage will be driven by a carefully planned education and outreach program that trains the user on the rideshare board, alleviates fears and concerns about safety, and encourages trust and faith in this reliable alternative.

#### References

- 4-traders. (2014, October 02). Verso Paper: Mill in Bucksport to close by year's end, 570 employees to lose jobs. 4-Traders. Retrieved from http://www.4-traders.com/VERSO-PAPER-CORP-3023920/news/Verso-Paper--mill-in-Bucksport-to-close-by-years-end-570employees-to-lose-jobs-19133732/.
- American Association of State Highway and Transportation Officials. (August 2006). Non-Metropolitan Local Consultation Process: A Self-Assessment Tool for States. Retrieved from http:// committeereview.transportation.org/Documents/AASHTO%20Org%20Charts

%202015%20Publications%201May15.pdf.

- American Commuter Survey Reports. (2009). *Commuting in the United States*. Retrieved from Retrieved from http://www.census.gov/prod/2011pubs/acs-15.pdf.
- Beecroft, M. & Pangbourne, K. (2014). Future prospects for personal security in travel by public transport. *Transportation Planning and Technology*, *38*(1), 131-149.
- Benmar, C. (2010). Hidden health costs of transportation. *Grist*. Retrieved from http://grist.org/article/2010-05-28-hidden-health-costs-of-transportation/.
- Bloomberg, L.D. & Volpe, M. (2012). *Completing your qualitative dissertation: A road map from beginning to end.* Thousand Oaks California: Sage Publications Inc.
- Boober, B.H. (2015). Keep Them Rocking at Home: Thriving in Place. *Maine Policy Review*, 24(2), 111-112.
- Brown, D.M. & Stommes, E.S. (2004). Rural Governments Face Public Transportation Challenges and Opportunities. *Amber Waves*, 2(1).
- Brown, A. (2013). Transit and the Upper Valley TMA. Tri-State Transit Conference, Burlington VT.

- Camoin Associates. (2015). *Supply chain analysis*. Retrieved from http://emdc.org/image..upload/Draft%20Report%205.18.2015.
- Campus Area Transportation Management Association (CATMA); Tri-State Transit Conference. September 13, 2013, Burlington, Vermont.
- Colgan, C.S. (September 2006). *Maine's Aging Economy and the Economy of Aging*. BlaineHouse Conference on Aging. Muskie School of Public Service, University of SouthernMaine.
- Committee on Sustainability Linkages in the Federal Government; Science and Technology for Sustainability Program (<u>STSP</u>); Policy and Global Affairs (<u>PGA</u>); National Research Council. (2013). Sustainability for the nation: Resource connection and governance linkages. *National Academies' Division on Policy and Global Affairs (PGA)*, Transportation Research Board (TRB), www.trb.org/Policy/Blurbs/169333. The National Academies Press; Washington, D.C.
- Community Transportation Association of America. (March 2004). *Medicaid expansion and premium assistance: The importance of non-emergency medical transportation (NEMT) to coordinated care for chronically ill patients*. Retrieved from http://web1.ctaa.org/webmodules/webarticles/articlefiles/NEMTreportfinal.pdf.
- Corbin, J.M. & Strauss, A.L. (2008). *Basics of qualitative research: Techniques and* procedures for developing grounded theory. Thousand Oaks: Sage Publications, Inc.
- Corless, J. (2011). More options are a must. *National Journal*. Retrieved from http://transportation.nationaljournal.com/2011/03/the-impact-of-high-gas-prices.php#1915479.

- Creswell, J.W. (2013). *Qualitative inquiry & research design: Choosing among five approaches*. Thousand Oaks: Sage Publications, Inc.
- Cromartie, J. (2013). How is Rural America Changing? United States Department of Agriculture, Economic Research Service. Retrieved from http://www.census.gov/newsroom/cspan/rural\_america /20130524\_rural\_america\_slides.pdf
- Deka, D. (2014). The role of household members in transporting adults with disabilities in the United States. *Transportation Research Part A*, 69, 45-57.
- Deloitte. (2007). Closing America's infrastructure gap: The role of public private partnerships. Deloitte Development LLC.
- Eastern Area Agency on Aging. Interview with Executive Director Noelle Merrill, April 22, 2015.
- Eastern Maine Development Corporation, Re-Engineering the Region. (2015). Short-Term
  Strategic Planning to Enhance Regional Recovery Following Rail Disaster and Major
  Industry-Targeted Layoffs Diversifying Economic and Transportation Options. Final
  Report. Retrieved from http://www.emdc.org.
- Eysenbach, G. (2001). What is ehealth? Journal of Medical Internet Research, 3(2), e20.
- Ferguson, E. (2007). Transportation management associations: A reappraisal. *Journal of Public Transportation*, 10 (4).
- Ferguson, E., Ross, C., Meyer, M. (1992). Transportation management associations:
  Organization, implementation, and evaluation. *Transportation Research Record 1346*,
  TRB (www.trb.org), 36-43.

- Fishell, D. (2014) Economists say Maine's aging population hinders growth. *Bangor Daily News*, 05/27/2014. Retrieved from http://bangordailynews.com/2014/06/26/business/ourworkforce-is-barely-growing-economists-say-maines-aging-population-hinders-growth/.
- Fletcher, C.N., Garasky, S.B., Jensen, H.H. & Nielsen, R.B. (2010). Transportation access: A key employment barrier for rural low-income families. *Journal of Poverty*, 14(2), 123-144.
- Gallup-Healthways Solutions. (2011). U.S. seniors maintain happiness highs with less social time. Retrieved from http://www.gallup.com/poll/151457/Seniors-Maintain-Happiness-Highs-Less-Social-Time.aspx.
- Glazer, B.G. & Strauss, A.L. (1967). *The discovery of grounded theory: Strategies for qualitative research*. Chicago: Aldine Publishing Company.
- Goins, R.T., Williams, K.A., Carter, M.W., Spencer, S.M. & Solovieva, T. (2005). Perceived barriers to health care access among rural older adults: A qualitative study. *The Journal* of Rural Health, 21(3), 206-213.
- Harvard Women's Health Watch. (March 2012). *Why behavior change is hard and why you should keep trying*, *19*(7), 4-5.
- Hawkins, A.J. (2015). Uber covers 75 percent of the US, but getting to 100 will be really hard. Retrieved from http://www.theverge.com/2015/10/23/9603324/uber-coverage-ruralareas-dominance-plan.
- Hill, C.B. (2015). Income inequality and higher education. American Council on Education, Leadership and Advocacy. Retrieved from http://www.acenet.edu/thepresidency/columns-and-features/Pages/Income-Inequality-and-Higher-Education.aspx.

- Hiskey, S. & McPherson, S. (2013). That's just life: older adult constructs of trauma. *Aging & Mental Health*, *17*(6), 689-696.
- Johnson, F. (2011). The impact of high gas prices. *National Journal*. Retrieved from http://transportation.nationaljournal.com/2011/03/the-impact-of-high-gas-prices.php#1915479.
- Kolodinsky, J.M., DeSisto, T.P., Propen, D., Putnam, M.E., Roche, E., & Sawyer, W.R. (2013).It is not how far you go, it is whether you can get there: Modeling the effects of mobility on quality of life in rural New England. *Journal of Transport Geography*, 31, 113-122.
- Eastern Maine Development Corporation. (2014). Linking the Rural Regions of Four Counties in Maine to Enhance Transportation Opportunities and Improve Quality of Life: TIGER2 Planning Grant Report. Retrieved from

http://www.emdc.org/document\_upload/TIGER\_II\_REPORT.pdf.

- Litman, T. (2005/2006). Transportation solutions. Race, poverty & the environment; Moving the movement: Transportation justice: 50 years after the Montgomery bus boycott, 12(1), 70-72.
- Litman, T. (2014). Win-Win Transportation Solutions: Mobility management strategies that provide economic, social and environmental benefits. Retrieved from http://vtpi.org/winwin.pdf.
- Littell, J.H. & Girvin, H. (2002). Stages of change: a critique. *Behavior Modification*, 26(2), 223-273.
- Loveless, B. (2015). Former oil exec: \$5-a-gallon gas on the way. Retrieved from http://www.usatoday.com/story/money/2015/01/19/gas-oil-five-dollar-gallon/21865975/.

- Loveless, S.M. & Welch, J.S. (1999). Growing to meet the challenges; emerging roles for transportation management associations. *Transportation Research Record 1659*, TRB (www.trb.org), 121-128.
- Maine Center for Disease Control & Prevention. *Healthy Maine 2010*. Retrieved from http://www.maine.gov/dhhs/mecdc/healthy-maine/index.shtml.

Maine Department of Transportation. (2010). High priority objectives and unmet needs.
 *Connecting Maine: Maine's Long Range Transportation Plan - Final Report, 27-30.* Retrieved from

http://www.maine.gov/mdot/connectingmaine/documents/pdf/Chapter3.pdf.

Maine Department of Transportation, Bureau of Planning. (2013). *MaineDOT Region 3, Locally Coordinated Plan, 3*. Retrieved from www.maine.gov/mdot/planning/docs/ptp/Region3Overview.pdf.

- Maine Department of Transportation. (2015) *Final Strategic Transit Plan 2025: Transforming Public Transit, Meeting Future Needs, Managing Expectations and Resources.* Retrieved from http://www.maine.gov/mdot/planning/docs/FInalStrategicPlan.pdf.
- Maine Office of Elder Services, Department of Health and Human Services. (2012). *State Plan* on Aging October 1, 2012 – September 30, 2016. Retrieved from http://digitalmaine.com/oads\_docs/2/.
- Maine State Housing Authority. (2008). A report on housing costs in Maine. Retrieved from http://www.mainehousing.org/docs/default-source/housing-reports/a-report-on-housingcosts-in-maine---2008.pdf?sfvrsn=4.
- McCray, T. (2001). Promoting the journey to health: Healthcare access and transportation in rural South Africa. *Urban Technological and Environment Planning*. University of Michigan.

- Mirriam, S.B. (2014). *Qualitative research: A guide to design and implementation.* 3<sup>rd</sup> Edition. Hoboken: Wiley.
- Mitton, C., Dionne, F., Masucci, L., Wong, S., & Laws, S. (2010). Innovations in health service organization and delivery in northern rural and remote regions: A review of the literature. *International Journal of Circumpolar Health*,70 (5).
- Morris, M.H. (September 2008). How to take adoption of transportation systems management and operations to the next level. *ITE Journal*, 19-22.
- National Center For Transit Research. *Rural Transit Fact Book 2014*. Retrieved from http://www.surtc.org/transitfactbook/downloads/2014-rural-transit-fact-book.pdf.
- National Household Transit Survey. (2009). Retrieved from http://nhts.ornl.gov/2009/pub/ UsersGuideV2.pdf.
- O'Neill, M., Woods, P., & Webster, M. (2005). New arrivals: Participatory action research, imagined communities, and "visions" of social justice. *Social Justice*, *32*(1), (99), 75-88.
- Partners for Livable Communities. (2015). *Aging in Place Initiative*. Retrieved from http://www.livable.org/program-areas/livable-communities-for-all-ages-a-aging-inplace/overview.
- Portland Plan. (2010). *Action Area: Transportation, technology and access*. Retrieved from http://www.portlandonline.com/portlandplan/index.cfm?a=297202&c=51314.
- Popuri, Y., Proussaloglou, K., Ayvalik, C., Koppelman, F. & Lee, A. (2011). Importance of traveler attitudes in the choice of public transportation to work: Findings from the Regional Transportation Authority Attitudinal Survey. *Transportation*, 38, 643-661.
- Rangarajan, K., Ziemer, N., & Long, S. (2007). Public private partnerships and rural economic development. Proceedings of the 2009 Industrial Engineering Research Conference.

Robert Wood Johnson Foundation: Health Policy Snapshot. 2012). How does transportation impact health? *Public Health and Prevention*. Retrieved from http://www.rwjf.org/healthpolicy.

- Rodrigue, J.P. & Notteboom, T. (2013). Transportation and economic development. In J.P. Rodrigue (Ed), *The Geography of Transport Systems, Third Edition*. New York, Routledge,
- Saldana, J., Leavy, P. & Beretvas, N. (2013). *Fundamentals of Qualitative Research*. New York: Oxford University Press, p. 95-102.
- Schwarzbach, A. R. (2011). High gas prices may drive behavior changes. *Arcata: Northcoast Environmental Center.* Retrieved from

http://search.proquest.com/docview/1265767680?accountid=12756

- Sharpiro, D., Pearlmutter, D., and Schwartz, M. (2012). The emergence of rural transport strategies in response to rising fuel costs. *Energy Policy*, 44, 92-100.
- Shergold, I., Parkhurst, G. & Musselwhite, C. (2012). Rural car dependence: An emerging barrier to community activity for older people. *Transportation Planning and Technology*, 35(1), 69-85.
- Simmons, A. (2013). Transportation: AR focuses on commuting alternatives: Plan called a good way to raise public awareness. 82% still choosing to drive alone. *The Atlanta Journal Constitution*, B.1.
- Srinivasan, V. (2015). *Docs on wheels: Mobile clinics are a great way to deliver health care.* Retrieved from

http://www.slate.com/articles/health\_and\_science/medical\_examiner/2015/05/mobile\_he alth\_clinics\_outcomes\_community\_relationships\_privacy\_concerns.html.

- Speedmatters. (2008). Lack of high-speed Internet puts rural areas at a disadvantage. Retrieved from http://www.speedmatters.org/blog/archive/lack\_of\_high\_speed\_ internet\_puts\_rural\_areas\_at\_a\_disadvantage.
- Steeves, H. (2008). Zipcar program revoked after students "voted with their feet". Maine Center for Disease Control & Prevention. *The Maine Campus*. Retrieved from http://mainecampus.com/2008/09/11/zipcar-program-revoked-after-students-voted-withtheir-feet/.
- Transportation Management Associations. (2011). *TDM Encyclopedia, Victoria Transport Policy Institute*. Retrieved from http://www.vtpi.org/tdm/tdm44.htm.
- TRIP: a national transportation research group. (October 2012). Maine transportation by the numbers: Meeting the state's need for safe and efficient mobility. Retrieved from http://www.tripnet.org/docs/ME\_Transportation\_by\_the\_Numbers\_TRIP\_Report\_Oct\_20 12.pdf.
- United States 2010 Census. *State & County QuickFacts*. Retrieved from http://quickfacts.census.gov/qfd/states/23000.html.
- United States 2012 Census. *State Totals Vintage 2012*. Retrieved from http://www.census.gov/popest/data/state/totals/2012/index.html.
- United States Department of Transportation. (2012). *Planning for transportation in rural areas*. Retrieved from http://www.fhwa.dot.gov/planning/publications/ rural\_areas\_planning/page03.cfm.
- United States Department of Transportation. (2011). *Federal Highway Administration: Highway Statistics 2011*. Retrieved from http://www.fhwa.dot.gov.

- United States Department of Transportation, Federal Transit Administration. MAP 21. (2014). *Moving ahead for progress in the 21<sup>st</sup> Century: Transforming the way we build, manage, and maintain our nation's transit systems.* Retrieved from http://www.fta.dot.gov/ map21/index.html.
- United States Energy Information Administration. (2016). *Short-term energy outlook*. Retrieved from http://www.eia.gov/forecasts/steo/.
- United Way of Eastern Maine. (2004). *Senior health transportation initiative*. Retrieved from http://www.unitedwayem.org/content/4059/senior-health-transportation.
- University of Maine Center on Aging. (2005). *The Eastern Maine Transportation Collaborative's* (*EMTC*) *health services initiative needs assessment research final report*. Retrieved from http://mainecenteronaging.umaine.edu/programs/research/transportation.
- Upper Valley Rideshare: *Green choices for commuters*. Retrieved from http://www.uppervalleyrideshare.com/.
- Urban Design 4 Health, Inc., American Public Health Association. (2012). *The hidden health costs of transportation*. Retrieved from http://www.apha.org/NR/rdonlyres/F84640FD-13CF-47EA-8267-E767A1099239/0/HiddenHealthCostsofTransportationShortFinal.pdf.
- Victoria Transport Policy Institute (VTPI). (2014). *Online Transportation Demand Management TDM) Encyclopedia*. Retrieved from http://www.vtpi.org/tdm/index.php.
- Vrabec, N.J. (1995). Implications of U.S. health care reform for the rural Elderly. *Nursing Outlook*, *43*, 260-265.
- Zhou, J. (2012). An analysis of university employee car-sharers in Los Angeles. *Transportation Research Part D*, *17*, 588-591.